

# On the geometry of normal horospherical *G*-varieties of complexity one

Kevin Langlois, Ronan Terpereau

### ▶ To cite this version:

Kevin Langlois, Ronan Terpereau. On the geometry of normal horospherical G-varieties of complexity one. Journal of Lie Theory, 2016, 26 (1), pp.49-78. hal-01081134v2

## HAL Id: hal-01081134 https://hal.science/hal-01081134v2

Submitted on 23 Dec 2016  $\,$ 

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

#### ON THE GEOMETRY OF NORMAL HOROSPHERICAL G-VARIETIES OF COMPLEXITY ONE

#### KEVIN LANGLOIS AND RONAN TERPEREAU

ABSTRACT. Let G be a connected simply-connected reductive algebraic group. In this article, we consider the normal algebraic varieties equipped with a horospherical G-action such that the quotient of a G-stable open subset is a curve. Let X be such a G-variety. Using the combinatorial description of Timashev, we describe the class group of X by generators and relations and we give a representative of the canonical class. Moreover, we obtain a smoothness criterion for X and a criterion to determine whether the singularities of X are rational or log-terminal respectively.

#### Contents

Introduction	1
1. Preliminaries	4
1.1. Horospherical homogeneous spaces	4
1.2. Models, colors, and valuations	5
1.3. Colored polyhedral divisors	7
2. Main results	11
2.1. Parabolic induction and smoothness criteria	11
2.2. Decoloration morphism	16
2.3. Parametrization of the stable prime divisors	18
2.4. Canonical class and log-terminal singularities	23
References	27

#### INTRODUCTION

The varieties and the algebraic groups that we consider are defined over an algebraically closed field k of characteristic zero. Let G be a connected simply-connected reductive algebraic group and let  $B \subset G$  be a Borel subgroup. The aim of this article is to describe certain geometric properties of a family of G-varieties: the normal horospherical G-varieties of complexity one. Among other results, we obtain explicit criteria to characterize the singularities of these varieties, we describe their class group by generators and relations, and we give an explicit representative of their canonical class.

Recall that the *complexity* of a *G*-variety X is the transcendence degree of the field extension  $k(X)^B$  over k, where  $k(X)^B$  denotes the field of *B*-invariant rational functions on X; see [LV83, Vin86]. Since all the Borel subgroups of G are

conjugated, this notion does not depend on the choice of B. Geometrically, the complexity is the codimension of a general B-orbit. For instance, the G-varieties of complexity zero contain an open B-orbit. Those which are normal are called *spherical varieties*; see [Kno91, Pez10, Hur11, Tim11, Per14] for more information.

The study of the G-varieties of complexity one is the next step (after the spherical case) towards the classification of normal G-varieties. Many important examples of G-varieties of complexity one motivate this study:

- The normal *T*-varieties of complexity one, where *T* is a torus; see [KKMS73, Tim08] for a combinatorial description, [Lan15] for a generalization over an arbitrary field, [FZ03] for the case of surfaces, and [AH06, AHS08, AIPSV12] for higher complexity.
- The homogeneous spaces G/H of complexity one, where H is a connected reductive closed subgroup, are described in [Pan92, AC04].
- The normal three-dimensional affine SL<sub>2</sub>-varieties with a two-dimensional general orbit are studied in [Arz97].
- The embeddings of  $SL_2/K$  into normal  $SL_2$ -varieties, where K is a finite subgroup, are studied in [Pop73], [LV83, §9], [Mos90], and [Mos92]. More generally, see [Tim11, §16.5] for a classification of the embeddings of G/K where G is a semisimple group of rank 1 and  $K \subset G$  is a finite subgroup.
- An example from classical geometry: Let  $T \subset SL_3$  be the subgroup of diagonal matrices. The homogeneous space  $SL_3/T$  can be identified with the set of ordered triangles on  $\mathbb{P}^2$ . The embeddings of  $SL_3/T$  are studied in [War82] and [Tim11, 16.5].

A combinatorial description of normal G-varieties of complexity one is obtained in [Tim97]. This description is inspired by the Luna–Vust theory of embeddings of homogeneous spaces G/H into normal varieties; see [LV83].

A G-action is called *horospherical* if the isotropy group of any point contains a maximal unipotent subgroup of G. Therefore a homogeneous space G/H is horospherical if and only if H contains a maximal unipotent subgroup; such an H is called a *horospherical subgroup* of G. It follows from the Bruhat decomposition of G that every horospherical G-homogeneous space is spherical, and thus a general G-orbit of a horospherical G-variety of complexity one has codimension one. We will recall in §1.1 how the set of horospherical subgroups of G.

The *G*-equivariant birational class of a horospherical *G*-variety *X* is determined by the invariant field  $k(X)^G$  and by the isotropy subgroup *H* of a general point. Indeed, if *r* denotes the complexity of the *G*-variety *X*, then by [Kno90, Satz 2.2] there exist an *r*-dimensional variety *C* and a *G*-equivariant birational map

$$\phi: X \to Z \coloneqq C \times G/H,$$

where G acts on  $Z = C \times G/H$  by translation on G/H. The map  $\phi$  induces field isomorphisms  $k(X) \simeq \operatorname{Frac}(k(C) \otimes_k k(G/H))$  and  $k(C) \simeq k(X)^G$ . If r = 1, then we can assume that C is a smooth projective curve. This article is structured as follows: In the first part, we set up our framework by explaining the combinatorial description of normal horospherical G-varieties of complexity one (following Timashev). Let us be more precise. In §1.1 we describe the horospherical G-homogeneous spaces. In §1.2 we recall the definition of the scheme of geometric localities  $\operatorname{Sch}_G(Z)$  whose normal separated G-stable open subsets of finite type, also called G-models of Z, are the normal G-varieties G-birational to Z. We also introduce the notion of a chart, which is a B-stable affine open subset of  $\operatorname{Sch}_G(Z)$ , and of a germ, which is a (proper) G-stable closed subvariety of  $\operatorname{Sch}_G(Z)$ . Then we consider the B-stable divisors on G/H, also called colors (of G/H), and we introduce the set of G-valuations of k(Z). Finally, we define the colored  $\sigma$ -polyhedral divisors in §1.3 and explain how to obtain any G-model of Zfrom a finite collection of such polyhedral divisors.

The results of this paper are stated and proved in the second part. In §2.1 we explain how to obtain any simple G-model of Z as the parabolic induction of an affine L-variety, where  $L \subset G$  is a Levi subgroup. In particular, we obtain an effective construction of any simple G-model of Z. From this, we deduce several criteria to characterize the singularities of simple G-models of Z; see Theorem 2.4 for a criterion for rationality of singularities and Theorems 2.5 and 2.6 for smoothness criteria. As mentioned at the end of §2.1, our smoothness criteria are explicit thanks to the works of Pauer, Pasquier and Batyrev–Moreau. In §2.2 we prove the existence of the decoloration morphism for normal horospherical G-varieties of complexity one and give an explicit description of this morphism in terms of germs; see Proposition 2.9 for a precise statement. The decoloration morphism was introduced by Brion for spherical varieties in [Bri91, §3.3] and plays a key-role in our study of normal horospherical G-varieties of complexity one. Until the end of this introduction, we let X be such a variety. In  $\S2.3$ , the heart of this paper, we parametrize the G-stable prime Weil divisors of X and deduce from this a description of the class group of Xby generators and relations; see Theorem 2.11 and Corollary 2.12. Then we obtain a criterion of factoriality for X; see Corollary 2.14. Also, we relate the description of stable Cartier divisors obtained by Timashev in [Tim00] to our description of stable Weil divisors; see Corollary 2.17. In §2.4 we give an explicit representative of the canonical class of X; see Theorem 2.18. From this, we deduce criteria for Xto be Q-Gorenstein or log-terminal respectively; see Corollary 2.19 and Theorem 2.22. Finally, Proposition 2.21 provides an explicit resolution of singularities of Xwhich factors through the decoloration morphism defined in §2.2.

Acknowledgements. Both authors are grateful to Michel Brion, Ariyan Javanpeykar, Boris Pasquier, and especially to the referee for their reading and their valuable comments. The authors express their gratitude to the Institute of Mathematical Sciences (ICMAT) of Madrid for the hospitality it provided during the writing of this paper. The first-named author benefited from the support of the ERC Consolidator Grant NMST. He also thanks the Max Planck Institut für Mathematik Bonn for support. The second-named author benefited from the support of the DFG via the SFB/TR 45 "Periods, Moduli Spaces and Arithmetic of Algebraic Varieties" and from the support of the GDR "Théorie de Lie Algébrique et Géométrique".

**Notation.** The base field k is algebraically closed of characteristic zero. An integral separated scheme of finite type over k is called a *variety*. If X is a variety, then k[X] denotes the coordinate ring of X and k(X) denotes the field of rational functions of X. A point of X is always assumed to be closed. We denote by G a connected simply-connected reductive algebraic group (i.e., a direct product of a torus and a connected simply-connected semisimple group), by  $B \subset G$  a Borel subgroup, by  $U = R_u(B)$  the unipotent radical of B, and by  $T \subset B$  a maximal (algebraic) torus. We denote by  $\mathbb{G}_m$  the multiplicative group over k. A subgroup of G is always a closed subgroup. If H is such a subgroup, then

$$N_G(H) = \{g \in G \mid gHg^{-1} \subset H\}$$

is the *normalizer* of H in G. For an algebraic group K, we denote by

 $\chi(K) = \{ \text{algebraic group homomorphisms } \phi : K \to \mathbb{G}_m \}$ 

the character group of K. A variety on which K acts (algebraically) is called a Kvariety. An algebra (over k) on which K acts by algebra automorphisms is called a K-algebra. If X is a K-variety, then k[X] and k(X) are K-algebras; in particular, k[X] and k(X) are linear representations of K.

#### 1. Preliminaries

In this first part, we explain the combinatorial description of normal G-varieties of complexity one as given in [Tim11, §16] and specialized in the horospherical case. Let C be a smooth projective curve, let G/H be a horospherical G-homogeneous space, and let  $Z = C \times G/H$ ; the group G acts on Z by translation on the second factor. The approach of Timashev consists in giving a classification of all the normal G-varieties which are G-birational to Z.

Several times in this first part (and also later on) we mention the reference of a result (for instance in [Tim11]) to justify an assertion or to introduce a notion but without recalling explicitly the result itself; the reason being that we did not wish to make this article too long and too technical. However, we tried to make it self-contained in the sense that whenever we believed that a notion or a result could be enlightening, then we recalled it explicitly.

1.1. Horospherical homogeneous spaces. In this section we enunciate a combinatorial description of the horospherical homogeneous spaces; see [Pas08, §2] for details.

Let S be the set of simple roots of G with respect to (T, B). There exists a well-known one-to-one correspondence  $I \mapsto P_I$  between the powerset of S and the set of parabolic subgroups of G containing B; see [Spr98, Th. 8.4.3]. Let us assume that the closed subgroup  $H \subset G$  contains the unipotent radical U of B. Then  $P = N_G(H)$  is a parabolic subgroup containing B. Therefore, there exists a unique subset  $I \subset S$  such that  $P = P_I$ . The quotient algebraic group  $K \coloneqq P/H$  is a torus and  $M = \chi(K)$  identifies naturally with a sublattice of  $\chi(T)$ .

The next statement ([Pas08, Prop. 2.4]) explains how the pair (M, I) completely describes the horospherical homogeneous space G/H.

#### **Proposition 1.1.** The map $G/H \mapsto (M, I)$ is a bijection between

- the set of closed subgroups of G containing U; and
- the set of pairs (M, I), where M is a sublattice of  $\chi(T)$  and I is a subset of S such that for every  $\alpha \in I$  and every  $m \in M$ , we have  $\langle m, \hat{\alpha} \rangle = 0$  (here  $\hat{\alpha}$  denotes the coroot of  $\alpha$ ).

1.2. Models, colors, and valuations. From now on, H is a closed subgroup of G associated with a pair (M, I) as in Proposition 1.1 and  $P = P_I$  is the parabolic subgroup  $N_G(H)$ . Also, we recall that  $Z = C \times G/H$ .

1.2.1. We first introduce the scheme of geometric localities as in [Tim11, §12.2]. All varieties which are birational to Z may be glued together into a scheme over k that we denote by Sch(Z). More precisely, the schematic points of Sch(Z) are local rings corresponding to prime ideals of finitely generated subalgebras with quotient field k(Z) and the spectra of those subalgebras define a base of the Zariski topology on Sch(Z) (by identifying prime ideals with associated local rings). Furthermore, the abstract group G acts on the set Sch(Z) via its linear action on k(Z). We denote by  $Sch_G(Z)$  the maximal normal open subscheme on which the action of G on Sch(Z) is regular; see [Tim11, Prop. 12.2]. A G-model of Z is a G-stable dense open subset of  $Sch_G(Z)$  which is separated and Noetherian (or equivalently separated and of finite type over k).

We now introduce the notions of chart and germ of a *G*-model X of Z. A chart (or affine chart or *B*-chart) of X is an affine dense open subset of X which is *B*-stable. A germ (or a *G*-germ) of X is a non-empty *G*-stable irreducible closed subvariety  $\Gamma \subsetneq X$ . By [Sum74, Th. 1], for every germ  $\Gamma \subsetneq X$  there exists a chart  $X_0 \subset X$  such that  $X_0 \cap \Gamma \neq \emptyset$ . The *G*-model X of Z is called simple if it has a chart intersecting all the germs. By [Tim00, §5, Lemma 2], every simple *G*-model of Z is quasi-projective. Moreover, X is a finite union of simple *G*-models of Z.

1.2.2. We now introduce the notion of color. Let K' be an algebraic group acting on a variety X', then a K'-divisor on X' is an irreducible closed subvariety of X'which is K'-stable and has codimension one. A color of G/H is a B-divisor on G/H. Let us consider the natural map

$$\pi: G/H \to G/P.$$

Then each color (of G/H) is of the form  $D_{\alpha} = \pi^{-1}(E_{\alpha})$ , where  $E_{\alpha}$  is the Schubert variety of codimension one corresponding to the root  $\alpha \in S \setminus I$ .

We represent colors as vectors of the lattice  $N = \text{Hom}_{\mathbb{Z}}(M,\mathbb{Z})$  as follows: For the natural action of B on k(G/H), the lattice  $M = \chi(P/H)$  identifies with the lattice of B-weights of the B-algebra k(G/H). For every (non-zero) B-eigenvector  $f \in k(G/H)$  of weight  $m \in M$ , we put

$$\langle m, \varrho(D_{\alpha}) \rangle = v_{D_{\alpha}}(f),$$

where  $v_{D_{\alpha}}$  is the valuation associated with  $D_{\alpha}$ . The value  $\rho(D_{\alpha})$  does not depend on the choice of f and coincides with the restriction of the coroot  $\hat{\alpha}$  to the lattice M; see [Pas08, §2]. Denoting by  $\mathcal{F}_0$  the set of colors, we obtain a map  $\rho: \mathcal{F}_0 \to N$ . Let us note that  $\rho$  is not injective in general; for instance, if H = P is a parabolic subgroup, then  $N = \{0\}$  and thus  $\rho$  is constant.

Let X be a G-model of Z. A B-divisor of X which is not G-stable is called a color of X. There is a one-to-one correspondence between the set of colors of G/H and the set of colors of X given as follows: X possesses a G-stable open subset of the form  $C' \times G/H$ , where  $C' \subset C$  is a dense open subset. If D is a color of G/H, then the closure of  $C' \times D$  in X is a color of X and vice-versa. In the following, we will always denote the colors of X and G/H in the same way.

1.2.3. A *G*-valuation of k(Z) is a function  $v: k(Z)^* \to \mathbb{Q}$  such that:

- $v(a+b) \ge \min\{v(a), v(b)\}$ , for all  $a, b \in k(Z)^*$  satisfying  $a+b \in k(Z)^*$ ;
- v is a group homomorphism from (k(Z)<sup>\*</sup>,×) to (Q,+) whose image is a discrete subgroup of (Q,+);
- the subgroup  $k^*$  is contained in the kernel of v; and
- $v(g \cdot a) = v(a)$  for every  $g \in G$  and every  $a \in k(Z)^*$ .

By [Tim11, Prop. 19.8], every *G*-valuation of k(Z) is proportional to a valuation  $v_D$  for a *G*-divisor *D* on a *G*-model of *Z*. Let us denote by  $N_{\mathbb{Q}} = \mathbb{Q} \otimes_{\mathbb{Z}} N$  the  $\mathbb{Q}$ -vector space associated with *N*. We follow [Tim11, Def. 16.1] and define the set  $\mathscr{E}$  as the disjoint union of sets  $\{z\} \times \mathscr{E}_z$ , where  $\mathscr{E}_z = N_{\mathbb{Q}} \times \mathbb{Q}_{\geq 0}$  and  $z \in C$ , modulo the equivalence relation ~ defined by

(1) 
$$(z, v, l) \sim (z', v', l')$$
 if and only if  $z = z', v = v', l = l'$  or  $v = v', l = l' = 0$ .

Therefore,  $\mathscr{E}$  is the disjoint union indexed by C of copies of the upper half-space  $N_{\mathbb{Q}} \times \mathbb{Q}_{\geq 0} \subset N_{\mathbb{Q}} \times \mathbb{Q}$  with boundaries  $N_{\mathbb{Q}} \times \{0\}$  identified as a common part.

There is a bijection between the set of G-valuations of k(Z) and the set  $\mathscr{E}$ , which we now explain. Let  $A_M$  denote the algebra generated by the B-eigenvectors of k(Z). Since M is a free abelian group, the exact sequence of abelian groups

$$0 \to \left(k(Z)^B\right)^* \to \left(k(Z)^{(B)}\right)^* \to M \to 0$$

splits. Let us fix once and for all a (non-canonical) splitting  $M \to (k(Z)^{(B)})^*$ ,  $m \mapsto \chi^m$ . Then  $A_M$  admits an *M*-grading given by

(2) 
$$A_M = \bigoplus_{m \in M} k(C) \chi^m$$

Let  $u = [(z, v, l)] \in \mathscr{E}$ . We define a valuation  $w = w_u$  of  $A_M$  as follows:

$$w\left(\sum_{i\in I}f_i\chi^{m_i}\right) = \min_{i\in I}\left\{v(m_i) + l\cdot\operatorname{ord}_z(f_i)\right\},$$

where I is a finite set, the  $m_i$  are pairwise distinct elements of M, and each  $f_i$  belongs to  $k(C)^*$ . By [Tim11, Cor. 19.13, Th. 20.3 and 21.10], for every  $u \in \mathscr{E}$ , there exists a unique G-valuation of k(Z) such that the restriction to  $A_M$  is  $w_u$ .

From now on, we will always identify  $\mathscr{E}$  with the set of *G*-valuations of k(Z) and  $N_{\mathbb{Q}}$  as a part of  $\mathscr{E}$  via the (well-defined) map  $v \mapsto [(\cdot, v, 0)]$ .

1.3. Colored polyhedral divisors. Let  $M_{\mathbb{Q}} = \mathbb{Q} \otimes_{\mathbb{Z}} M$ , then the  $\mathbb{Q}$ -vector spaces  $M_{\mathbb{Q}}$  and  $N_{\mathbb{Q}}$  are dual to each other; we denote the duality by

$$M_{\mathbb{O}} \times N_{\mathbb{O}} \to \mathbb{Q}, \ (m, v) \mapsto \langle m, v \rangle$$

We recall that a strongly convex polyhedral cone in  $N_{\mathbb{Q}}$  is a cone generated by a finite number of vectors and which contains no line.

1.3.1. We now recall the notions of colored cone and colored hypercone since these notions will intervene several times in the following; see also [Tim11, §§15.1, 16.3, and 16.4] for more details.

**Definition 1.2.** A colored cone of G/H is a pair  $(\mathcal{C}, \mathcal{F})$ , where  $\mathcal{F} \subset \mathcal{F}_0$  is a set of colors of G/H such that  $0 \notin \varrho(\mathcal{F})$ , and  $\mathcal{C} \subset N_{\mathbb{Q}}$  is a strongly convex polyhedral cone generated by  $\varrho(\mathcal{F})$  and finitely many other vectors.

**Definition 1.3.** Let  $C_0$  be a dense open subset of C. A hypercone of G/H is a union  $\mathscr{C} = \bigcup_{z \in C_0} \{z\} \times \mathscr{C}_z$  of convex polyhedral cones  $\mathscr{C}_z \subset N_{\mathbb{Q}} \times \mathbb{Q}_{\geq 0}$  such that

- for all but finitely many  $z \in C_0$  we have  $\mathscr{C}_z = (\mathscr{C} \cap N_{\mathbb{Q}}) + \mathbb{Q}_{\geq 0}\epsilon$  with  $\epsilon = (0, \ldots, 0, 1) \in N_{\mathbb{Q}} \times \mathbb{Q}$ ; and
- either (A) there exists  $z \in C_0$  with  $\mathscr{C}_z = (\mathscr{C} \cap N_{\mathbb{Q}})$ , or
  - (B) we have  $\emptyset \neq \mathcal{B} \coloneqq \sum_{z \in C_0} \mathcal{B}_z \subset (\mathscr{C} \cap N_{\mathbb{Q}})$ , where  $\epsilon + \mathcal{B}_z = \mathscr{C}_z \cap (\epsilon + N_{\mathbb{Q}})$ .

We say that  $\mathscr{C}$  is *strongly convex* if all  $\mathscr{C}_z$  are strongly convex and if  $0 \notin \mathcal{B}$ .

A colored hypercone of G/H is a pair  $(\mathscr{C}, \mathcal{F})$ , where  $\mathcal{F} \subset \mathcal{F}_0$  is a set of colors of G/H such that  $0 \notin \varrho(\mathcal{F})$ , and  $\mathscr{C} \subset \mathscr{E}$  is a strongly convex hypercone whose each  $\mathscr{C}_z$  is generated by  $\varrho(\mathcal{F}) \times \{0\}$  and finitely many other vectors.

A hyperface of a colored hypercone  $(\mathscr{C}, \mathcal{F})$  is a colored hypercone  $(\mathscr{C}', \mathcal{F}')$ , where  $\mathscr{C}' = \bigcup_{z \in C_0} \mathscr{C}'_z$  is a union of faces of  $\mathscr{C}_z$ , and  $\mathcal{F}' = \mathcal{F} \cap \varrho^{-1}(\mathscr{C}')$ .

Also, we recall that there is a correspondence between the colored cones of a horospherical homogeneous space and its simple equivariant embeddings; see [Kno91, Th. 3.1] for details.

1.3.2. In this subsection we introduce the notion of colored polyhedral divisor. This notion is equivalent to the one of colored hypercone in  $\mathscr{E}$  defined above, but is more suitable for our purposes.

**Definition 1.4.** Let  $\sigma \subset N_{\mathbb{Q}}$  be a strongly convex polyhedral cone. A  $\sigma$ -polyhedron is a subset of  $N_{\mathbb{Q}}$  obtained as a Minkowski sum  $Q + \sigma$ , where  $Q \subset N_{\mathbb{Q}}$  is the convex hull of a non-empty finite subset. Let  $C_0$  be a dense open subset of the curve C, let

$$\mathfrak{D} = \sum_{z \in C_0} \Delta_z \cdot [z]$$

be a formal sum over the points of  $C_0$ , where each  $\Delta_z$  is a  $\sigma$ -polyhedron of  $N_{\mathbb{Q}}$  and  $\Delta_z = \sigma$  for all but a finite number of  $z \in C_0$ , and let  $\mathcal{F} \subset \mathcal{F}_0$  be a set of colors of G/H such that

- 0 does not belong to  $\rho(\mathcal{F})$ ; and
- $\varrho(\mathcal{F}) \subset \sigma$ .

We call such a pair  $(\mathfrak{D}, \mathcal{F})$  a colored  $\sigma$ -polyhedral divisor on  $C_0$ . If  $\sigma$  and  $\mathcal{F}$  are clear from the context, then we write  $\mathfrak{D}$  instead of  $(\mathfrak{D}, \mathcal{F})$  and call  $\mathfrak{D}$  a colored polyhedral divisor on  $C_0$ . We say that  $\mathfrak{D}$  is trivial (on  $C_0$ ) if  $\Delta_z = \sigma$  for every  $z \in C_0$ .

It is crucial to keep in mind that a colored  $\sigma$ -polyhedral divisor is defined on a dense open subset  $C_0 \subset C$  and not on C itself. In the following,  $\sigma \subset N_{\mathbb{Q}}$  always denotes a strongly convex polyhedral cone and  $\mathcal{F}$  a set of colors satisfying the conditions of Definition 1.4; in particular,  $(\sigma, \mathcal{F})$  is a colored cone of G/H.

Let

$$\sigma^{\vee} = \{ m \in M_{\mathbb{O}} \mid \forall v \in \sigma, \langle m, v \rangle \ge 0 \}$$

denote the dual polyhedral cone of  $\sigma$ , and let  $\mathfrak{D}$  be a colored  $\sigma$ -polyhedral divisor on  $C_0 \subset C$ . To each  $m \in \sigma^{\vee}$ , we associate a  $\mathbb{Q}$ -divisor on  $C_0$ :

(3) 
$$\mathfrak{D}(m) = \sum_{z \in C_0} \min_{v \in \Delta_z(0)} \langle m, v \rangle \cdot [z],$$

where for every  $z \in C_0$  we denote by  $\Delta_z(0)$  the set of vertices of  $\Delta_z$ .

To a given  $\mathfrak{D}$  on  $C_0$  we also associate the following *M*-graded normal *k*-algebra (see [AH06, §3] for details):

(4) 
$$A[C_0,\mathfrak{D}] \coloneqq \bigoplus_{m \in \sigma^{\vee} \cap M} A_m \chi^m$$

where

$$A_m = H^0(C_0, \mathcal{O}_{C_0}(\mathfrak{D}(m))) \coloneqq H^0(C_0, \mathcal{O}_{C_0}(\lfloor \mathfrak{D}(m) \rfloor))$$

and  $\lfloor \mathfrak{D}(m) \rfloor$  is the Weil divisor (with integer coefficients) on  $C_0$  obtained by taking the integer part of each coefficient of  $\mathfrak{D}(m)$ . The multiplication on  $A[C_0, \mathfrak{D}]$  is constructed from the maps

 $\tau_{m,m'}: A_m \times A_{m'} \to A_{m+m'}, \ (f_1, f_2) \mapsto f_1 \cdot f_2.$ 

Let us note that each map  $\tau_{m,m'}$  is well-defined since for all  $m, m' \in \sigma^{\vee}$ :

$$\mathfrak{D}(m) + \mathfrak{D}(m') \leq \mathfrak{D}(m+m').$$

We now introduce the localization of a colored  $\sigma$ -polyhedral divisor. We will use this notion in the proof of Theorem 2.11.

**Definition 1.5.** With the notation above, let  $w \in \sigma^{\vee} \cap M$  such that  $A_w \neq \{0\}$  and  $f \in A_w \setminus \{0\}$ , and let  $\mathcal{F}^w$  be the set of colors defined by the relation

$$\mathcal{F}^w = \{ D \in \mathcal{F} \mid \varrho(D) \in w^\perp \}.$$

The *localization* of the colored  $\sigma$ -polyhedral divisor  $(\mathfrak{D}, \mathcal{F})$  with respect to  $f\chi^w$  is the colored  $(\sigma \cap w^{\perp})$ -polyhedral divisor  $(\mathfrak{D}_f^w, \mathcal{F}^w)$  defined by

$$\mathfrak{D}_f^w = \sum_{z \in (C_0)_f^w} \operatorname{Face}(\Delta_z, w) \cdot [z],$$

where

$$(C_0)_f^w = C_0 \setminus Z(f)$$
 with  $Z(f) = \operatorname{Supp}(\operatorname{div} f + \mathfrak{D}(w))$ , and

9

Face
$$(\Delta_z, w) \coloneqq \left\{ v \in \Delta_z \mid \langle w, v \rangle \leq \min_{v' \in \Delta_z} \langle w, v' \rangle \right\}.$$

To ensure that the algebra  $A[C_0, \mathfrak{D}]$  is finitely generated over k and has  $\operatorname{Frac} A_M$ as field of fractions (we recall that  $A_M$  denotes the algebra generated by the *B*eigenvectors of k(Z)), we now introduce the notion of properness for colored polyhedral divisors following [AH06, §2].

**Definition 1.6.** Let  $\mathfrak{D}$  be a colored  $\sigma$ -polyhedral divisor on  $C_0$ . Then  $\mathfrak{D}$  is called *proper* if either  $C_0$  is affine or  $C_0 = C$  is projective and satisfies the following conditions:

- deg  $\mathfrak{D} \coloneqq \sum_{z \in C} \Delta_z \not\subseteq \sigma$ .
- If  $\min_{v \in \deg \mathfrak{D}} \langle m, v \rangle = 0$ , then  $r\mathfrak{D}(m)$  is a principal divisor for some  $r \in \mathbb{Z}_{>0}$ .

Let us note that, if  $\mathfrak{D}$  is a proper colored polyhedral divisor on  $C_0$ , then by [AHS08, Prop. 3.3] we have the relation

$$A[C_0,\mathfrak{D}]_{f\chi^w} = A[(C_0)^w_f,\mathfrak{D}^w_f].$$

The next remark makes the link between the notions introduced in this subsection and the notions introduced in §1.3.1.

Remark 1.7. Let  $(\mathfrak{D}, \mathcal{F})$  be a colored  $\sigma$ -polyhedral divisor on a dense open subset  $C_0 \subset C$ . Let  $\mathscr{C}(\mathfrak{D})$  be the subset of  $\mathscr{E}$  defined as the disjoint union  $\sqcup_{z \in C_0} \{z\} \times \mathscr{C}(\mathfrak{D})_z$ modulo the equivalence relation ~ defined by (1), where  $\mathscr{C}(\mathfrak{D})_z$  is the cone generated by  $\sigma \times \{0\}$  and  $\Delta_z \times \{1\}$ . Then the pair  $(\mathscr{C}(\mathfrak{D}), \mathcal{F})$  is the colored hypercone of G/Hassociated with  $(\mathfrak{D}, \mathcal{F})$ . One may check that this gives a one-to-one correspondence between the set of colored polyhedral divisors defined on a dense open subset  $C_0 \subset C$ and the set of colored hypercones of  $\mathscr{E}$ . Moreover, through this correspondence, the properness of colored polyhedral divisors corresponds to the *admissibility* of colored hypercones; see [Tim11, Def. 16.12].

1.3.3. In this subsection, we explain how to construct a simple G-model of Z (introduced in §1.2.1) starting from a proper colored polyhedral divisor; see [Tim11, §13] for details.

Let  $(\mathfrak{D}, \mathcal{F})$  be a proper colored  $\sigma$ -polyhedral divisor on a dense open subset  $C_0 \subset C$ . Let us denote by  $\mathscr{C}(\mathfrak{D})(1)$  the set of elements  $[(z, v, l)] \in \mathscr{C}(\mathfrak{D})$  such that (v, l) is the primitive vector of an extremal ray of  $\mathscr{C}(\mathfrak{D})_z$ , and let  $Bx_0$  be the open B-orbit of G/H. Let us consider the subalgebra  $A \subset k(Z)$  defined by:

$$A = (k(C) \otimes_k k[Bx_0]) \cap \bigcap_{D \in \mathcal{F}} \mathcal{O}_{v_D} \cap \bigcap_{v \in \mathscr{C}(\mathfrak{D})(1)} \mathcal{O}_v,$$

where for a discrete G-valuation v of k(Z),

$$\mathcal{O}_{v} = \{ f \in k(Z)^{*} \mid v(f) \ge 0 \} \cup \{ 0 \}$$

is the corresponding local ring. By [Tim11, Th. 13.8, [16.4], A is a normal algebra of finite type over k. Moreover, by [Tim11, Cor. 13.9, [16.4], the affine B-variety

$$X_0(\mathfrak{D}) \coloneqq \operatorname{Spec} A$$

is an open subset of  $\operatorname{Sch}_G(Z)$ . Also, by [Tim11, Th. 12.6], the subscheme

$$X(\mathfrak{D}) \coloneqq G \cdot X_0(\mathfrak{D})$$

is a *G*-model of *Z* and  $X_0(\mathfrak{D})$  is a chart of  $X(\mathfrak{D})$ . In particular,  $X_0(\mathfrak{D})$  is a dense open subset of  $X(\mathfrak{D})$ . Conversely, for every simple *G*-model *X* of *Z* there exists a colored polyhedral divisor  $\mathfrak{D}$  such that  $X = X(\mathfrak{D})$ ; see [Tim11, Th. 16.19].

**Lemma 1.8.** Let  $(\mathfrak{D}, \mathcal{F})$  be a proper colored  $\sigma$ -polyhedral divisor on a dense open subset  $C_0 \subset C$ , and let  $X_0(\mathfrak{D}) \subset X(\mathfrak{D})$  be the corresponding chart. Then  $k[X_0(\mathfrak{D})]^U$ identifies with  $A[C_0, \mathfrak{D}]$  as K-algebras, where  $A[C_0, \mathfrak{D}]$  is the algebra defined by (4) and K is the torus introduced in §1.1.

Proof. The subalgebra

$$k[X_0(\mathfrak{D})]^U \subset (k(C) \otimes_k k[Bx_0])^U = A_M \text{ (defined in §1.2.3)}$$

is generated by elements of the form  $f\chi^m$ , where  $f \in k(C)^*$  and  $m \in M$ , satisfying  $v_D(f\chi^m) \ge 0$  and  $v(f\chi^m) \ge 0$  for all  $D \in \mathcal{F}$  and  $v \in \mathscr{C}(\mathfrak{D})(1)$ . One may check that these conditions are equivalent to  $\operatorname{div}(f) + \mathfrak{D}(m) \ge 0$ , which proves the lemma.  $\Box$ 

*Remark* 1.9. By [Tim97, Th. 3.1], and with the notation of Definition 1.5, we have the relation

$$X(\mathfrak{D}_f^w) = G \cdot (X_0(\mathfrak{D}))_{f\chi^w},$$

where  $(X_0(\mathfrak{D}))_{f\chi^w}$  is the distinguished Zariski open subset of  $f\chi^w \in k[X_0(\mathfrak{D})]$ .

1.3.4. Let X be the simple G-model of Z associated with the proper colored  $\sigma$ -polyhedral divisor  $(\mathfrak{D}, \mathcal{F})$  as explained in §1.3.3. By [Tim11, Th. 16.19 (2)], there is a combinatorial description of the set of germs of X (defined in §1.2.1). Each germ  $\Gamma$  is described by a pair  $(\mathcal{C}_1, \mathcal{F}_1)$ , called the *colored datum* of  $\Gamma$ , where  $\mathcal{C}_1$  is a hyperface of the hypercone  $\mathscr{C}(\mathfrak{D})$  (see Remark 1.7) and  $\mathcal{F}_1 \coloneqq \{D_\alpha \in \mathcal{F} \mid \varrho(D_\alpha) \in \mathcal{C}_1\}$ . Geometrically,  $\mathcal{F}_1$  is the subset of colors of  $\mathcal{F}$  that contain the germ  $\Gamma$ .

1.3.5. In this subsection we consider the general description of the non-necessarily simple G-models of Z.

A finite collection  $\Sigma = \{(\mathfrak{D}^i, \mathcal{F}^i)\}_{i \in J}$  of proper colored polyhedral divisors defined on dense open subsets of C is a *colored divisorial fan* if for all  $i, j \in J$  there exists  $l \in J$  such that  $\mathscr{C}(\mathfrak{D}^l) = \mathscr{C}(\mathfrak{D}^i) \cap \mathscr{C}(\mathfrak{D}^j)$  and  $(\mathscr{C}(\mathfrak{D}^l), \mathcal{F}^l)$  is a common hyperface of the colored hypercones  $(\mathcal{C}(\mathfrak{D}^i), \mathcal{F}^i)$  and  $(\mathcal{C}(\mathfrak{D}^j), \mathcal{F}^j)$ .

Let us denote

$$|\Sigma| = \bigcup_{i \in J} \mathscr{C}(\mathfrak{D}^i) \subset \mathscr{E}$$

the support of the colored divisorial fan  $\Sigma$ .

The next theorem gives a description of the *G*-models of Z; see [Tim11, Th. 16.19 (3)] and [Tim11, Cor. 12.14].

**Theorem 1.10.** If  $\Sigma = \{(\mathfrak{D}^i, \mathcal{F}^i)\}_{i \in J}$  is a colored divisorial fan on C, then the union  $X(\Sigma) \coloneqq \bigcup_{i \in J} X(\mathfrak{D}^i)$  in the scheme  $\operatorname{Sch}_G(Z)$  (defined in §1.2.1) is a G-model of Z, and every G-model of Z is obtained in this way. Moreover, the G-model  $X(\Sigma)$  of Z is a complete variety if and only if  $|\Sigma| = \mathscr{E}$ , where  $\mathscr{E}$  is the set of G-valuations (defined in §1.2.3).

#### 2. Main results

2.1. Parabolic induction and smoothness criteria. In this section we prove Lemma 2.1 which allows us to construct explicitly any simple *G*-model of *Z* as the parabolic induction of an affine *L*-variety, where  $L \subset G$  is a Levi subgroup. From this we deduce a criterion (Theorem 2.4) to determine whether the singularities of a simple *G*-model of *Z* are rational. Moreover, we obtain smoothness criteria (Theorems 2.5 and 2.6) for simple *G*-models of *Z*. Throughout this section we fix a proper colored  $\sigma$ -polyhedral divisor  $(\mathfrak{D}, \mathcal{F})$  on a dense open subset  $C_0 \subset C$ .

2.1.1. This subsection is inspired by [Tim11, §28]. Let us now introduce the notation that we will need to describe a simple G-model as the parabolic induction of an affine L-variety. As before, T denotes a maximal torus of G contained in B. We consider the following set of simple roots

$$I' \coloneqq \{ \alpha \in S \smallsetminus I \mid D_{\alpha} \in \mathcal{F} \} \cup I$$

and we denote by  $P_{\mathcal{F}}$  the parabolic subgroup  $P_{I'}$  containing B. We choose a Levi subgroup  $L \subset P_{\mathcal{F}}$  containing T and a Borel subgroup  $B_L$  of L containing T such that I' is the set of simple roots of L (with respect to  $(T, B_L)$ ). To summarize, we have the following inclusions:

$$T \subset B_L \subset L \subset P_{\mathcal{F}} \subset G.$$

We denote by  $Z_L$  the horospherical *L*-variety  $C \times L/H_L$ , where  $H_L = L \cap H$ . By [Tim11, Cor. 15.6], the homogeneous space  $L/H_L$  is quasi-affine. Moreover, denoting  $P_L = N_L(H_L)$ , the torus  $P_L/H_L$  identifies with K = P/H, and  $M = \chi(P/H) = \chi(P_L/H_L)$  through this identification. Let  $\mathcal{F}_L$  be the set of colors of  $L/H_L$ , then one may check that the image of  $\mathcal{F}_L$  by the map  $\varrho$  in the lattice N is the same as the one of  $\mathcal{F}$ . In particular,  $\mathcal{F}_L$  satisfies the conditions of Definition 1.4. We denote by  $(\mathfrak{D}_L, \mathcal{F}_L)$  the colored  $\sigma$ -polyhedral divisor on  $C_0$  (relatively to  $Z_L$ ) defined by the formal sum  $\mathfrak{D}_L := \mathfrak{D} = \sum_{z \in C_0} \Delta_z \cdot [z]$ , and by  $X(\mathfrak{D}_L) \subset \operatorname{Sch}_L(Z_L)$  the associated L-variety; see §1.3.3.

The quotient morphism  $P_{\mathcal{F}} \to P_{\mathcal{F}}/R_u(P_{\mathcal{F}}) \cong L$  makes  $X(\mathfrak{D}_L)$  a  $P_{\mathcal{F}}$ -variety. As the quotient morphism  $G \to G/P_{\mathcal{F}}$  is locally trivial (for the Zariski topology), we can form the *twisted product*  $G \times^{P_{\mathcal{F}}} X(\mathfrak{D}_L)$ . The latter is defined as the quotient  $(G \times X(\mathfrak{D}_L))/P_{\mathcal{F}}$ , where  $P_{\mathcal{F}}$  acts as follows:

 $p.(g,x) \coloneqq (gp^{-1}, p.x), \text{ where } p \in P_{\mathcal{F}}, g \in G, \text{ and } x \in X(\mathfrak{D}_L).$ 

We recall that the twisted product  $G \times^{P_{\mathcal{F}}} X(\mathfrak{D}_L)$  is a locally trivial fiber bundle over  $G/P_{\mathcal{F}}$  with fiber  $X(\mathfrak{D}_L)$ ; see [Jan87, §I.5] for more details on twisted products.

2.1.2. The next result is an adaptation of [PV72, §3] and [Pau81, §5] to the case of complexity one.

**Lemma 2.1.** With the notation of §2.1.1, the L-variety  $X(\mathfrak{D}_L)$  is affine and  $X(\mathfrak{D})$  is G-isomorphic to the twisted product  $G \times^{P_{\mathcal{F}}} X(\mathfrak{D}_L)$ . Moreover, the L-algebra

 $k[X(\mathfrak{D}_L)]$  identifies with the L-subalgebra

$$\mathcal{A}[C_0,\mathfrak{D}_L] \coloneqq \bigoplus_{m \in \sigma^{\vee} \cap M} H^0(C_0,\mathcal{O}_{C_0}(\mathfrak{D}(m))) \otimes_k V(m) \subset k(C) \otimes_k k[L/H_L],$$

where  $\mathfrak{D}(m)$  is the  $\mathbb{Q}$ -divisor on  $C_0$  defined by (3) and V(m) is the simple Lsubmodule of  $k[L/H_L]$  of highest weight m with respect to  $(T, B_L)$ .

*Proof.* The *G*-isomorphism between  $G \times^{P_{\mathcal{F}}} X(\mathfrak{D}_L)$  and  $X(\mathfrak{D})$  is a straightforward consequence of [Tim11, Prop. 14.4 and 20.13]. As  $\mathcal{F}_L$  is the set of colors of  $L/H_L$ , the  $B_L$ -variety  $X(\mathfrak{D}_L)$  is the chart corresponding to  $\mathfrak{D}_L$  (see the remark before [Tim11, Cor. 13.10]), and thus  $X(\mathfrak{D}_L)$  is affine.

As  $X(\mathfrak{D}_L)$  is horospherical, we have the following *L*-algebra decomposition (see [Tim11, Prop. 7.6]):

$$k[X(\mathfrak{D}_L)] \simeq \bigoplus_{m \in \sigma^{\vee} \cap M} k[X(\mathfrak{D}_L)]_m^{(B_L)} \otimes_k V(m),$$

where  $k[X(\mathfrak{D}_L)]_m^{(B_L)}$  denotes the vector space generated by the  $B_L$ -eigenvectors of weight m on which L acts trivially. The vector space  $k[X(\mathfrak{D}_L)]_m^{(B_L)}$  identifies with the space of global sections of  $\lfloor \mathfrak{D}(m) \rfloor$ . This proves the lemma.

*Remark* 2.2. With the notation above, one may check that  $X(\mathfrak{D}_L) = X_0(\mathfrak{D}_L)$ .

**Example 2.3.** Let us assume that  $C_0 = C$  is projective. Let  $G = SL_2$  and let H be the unipotent radical of the subgroup of upper triangular matrices. Then  $M = N = \mathbb{Z}$  and the set of colors  $\mathcal{F}_0$  of G/H is a singleton. Let  $\sigma = \mathbb{Q}_{\geq 0}$  and let  $\mathfrak{D}$  be a proper  $\sigma$ -polyhedral divisor on C such that the Weil  $\mathbb{Q}$ -divisor  $\mathfrak{D}(1)$  on C is integral and very ample. Let us note that, in this situation, we have G = L and thus  $\mathfrak{D} = \mathfrak{D}_L$ . Then by Lemma 2.1, the G-variety  $X(\mathfrak{D}, \mathcal{F}_0)$  is affine and  $k[X(\mathfrak{D}, \mathcal{F}_0)]$  identifies with

$$\bigoplus_{d\geq 0} H^0(C, \mathcal{O}_C(d.\mathfrak{D}(1))) \otimes_k V(d)$$

as a *G*-algebra, where V(d) is the irreducible representation of *G* of dimension d + 1 obtained by linearizing the line bundle  $\mathcal{O}_{\mathbb{P}^1}(d)$ . Then the line bundle  $\mathscr{L} := \mathcal{O}_C(\mathfrak{D}(1)) \boxtimes \mathcal{O}_{\mathbb{P}^1}(1)$  on  $C \times \mathbb{P}^1$  is very ample, and the *G*-variety  $X(\mathfrak{D}, \mathcal{F}_0)$  can be realized as the affine cone over the *G*-equivariant embedding of  $C \times \mathbb{P}^1$  in the projectivization of the space of global sections of  $\mathscr{L}$ .

2.1.3. The next results are criteria to characterize the singularities of a simple G-model  $X(\mathfrak{D})$  of Z; see [LS13, Prop. 5.1] for the case of normal T-varieties and [Tim00, §6, Th. 7] for a criterion for rationality of singularities in the general setting of normal G-varieties of complexity one.

We recall that a normal variety X has rational singularities if there exists a resolution of singularities  $\phi : Y \to X$  such that the higher direct images of  $\phi_*$ applied to  $\mathcal{O}_Y$  vanish. This notion does not depend on the choice of the resolution of singularities.

**Theorem 2.4.** Let  $(\mathfrak{D}, \mathcal{F})$  be a proper colored  $\sigma$ -polyhedral divisor on  $C_0$ . The simple G-model  $X(\mathfrak{D})$  of Z has rational singularities if and only if one of the following assertions holds.

- (i) The curve  $C_0$  is affine.
- (ii) The curve  $C_0$  is the projective line  $\mathbb{P}^1$  and deg  $\lfloor \mathfrak{D}(m) \rfloor \ge -1$  for every  $m \in \sigma^{\vee} \cap M$ , where  $\mathfrak{D}(m)$  is the  $\mathbb{Q}$ -divisor on  $C_0$  defined by (3).

*Proof.* In this proof, we identify  $X(\mathfrak{D})$  with the parabolic induction of  $X(\mathfrak{D}_L)$  using Lemma 2.1. Let us denote by  $B\bar{x}_0$  the open *B*-orbit of  $G/P_{\mathcal{F}}$  and let

$$q: X(\mathfrak{D}) = G \times^{P_{\mathcal{F}}} X(\mathfrak{D}_L) \to G/P_{\mathcal{F}}$$

be the (*G*-equivariant) projection. Then  $q^{-1}(B\bar{x}_0) \simeq B\bar{x}_0 \times X(\mathfrak{D}_L)$  is a chart of  $X(\mathfrak{D})$  intersecting all the germs of  $X(\mathfrak{D})$ . As  $B\bar{x}_0$  is an affine space, it follows from [Tim11, Th. D5 (3)] that  $X(\mathfrak{D})$  has rational singularities if and only if  $X(\mathfrak{D}_L)//U_L \simeq \operatorname{Spec} A[C_0, \mathfrak{D}]$  has rational singularities. We conclude by the rationality criterion for torus actions given in [LS13, Prop. 5.1].

The next theorem gives a smoothness criterion for  $X(\mathfrak{D})$  when  $(\mathfrak{D}, \mathcal{F})$  is a proper colored polyhedral divisor on an affine curve  $C_0$ ; see [Bri91, §4.2] for the spherical case, [LS13b, Prop. 5.1 and Th. 5.3] for the case of normal *T*-varieties of complexity one, and [Mos90, §3] for the case of embeddings of SL<sub>2</sub> and PSL<sub>2</sub>. Our proof is inspired by a description of toroidal embeddings given in [KKMS73, §II]. Roughly speaking, we prove that the smoothness of  $X(\mathfrak{D})$ , which we recall is a normal horospherical *G*-variety of complexity 1, is tantamount to the smoothness of some normal horospherical  $\mathbb{G}_m \times G$ -varieties of complexity 0.

**Theorem 2.5.** With the same notation as before, and assuming that  $C_0$  is affine, the following statements are equivalent:

- (i) The G-variety  $X(\mathfrak{D})$  is smooth.
- (ii) For every  $z \in C_0$ , the simple embeddings of the  $\mathbb{G}_m \times G$ -homogeneous space  $\mathbb{G}_m \times G/H$  associated with the colored cones  $(\mathscr{C}(\mathfrak{D})_z, \mathcal{F})$  (see §1.3.1) are smooth.

Proof. Let us fix  $z \in C_0$ . As  $C_0$  is smooth, by [EGA IV, IV,17.11.4], there exist open subsets  $U_z \subset C_0$  and  $V_z \subset \mathbb{A}^1$  containing z and 0 respectively, and there exists an étale morphism  $\tau_z : U_z \to V_z$  such that  $\tau_z(z) = 0$ . Let  $\mathfrak{D}'_L \coloneqq \sum_{x \in \mathbb{A}^1} \Delta'_x \cdot [x]$  with  $\Delta'_0 = \Delta_z$  and  $\Delta'_x = \sigma$  for all  $x \neq 0$  be a polyhedral divisor on  $\mathbb{A}^1$ . Let  $\mathfrak{D}_{L|U_z}$  and  $\mathfrak{D}'_{L|V_z}$  denote the polyhedral divisor obtained by restricting  $\mathfrak{D}_L$  and  $\mathfrak{D}'_L$  on  $U_z$  and  $V_z$  respectively. One may check that the morphism  $\tau_z$  induces an étale morphism of algebras  $\phi_z : \mathcal{A}[V_z, \mathfrak{D}'_{L|V_z}] \to \mathcal{A}[U_z, \mathfrak{D}_{L|U_z}]$ . By Lemma 2.1, the morphism  $\phi_z$  in turn induces an étale morphism  $\delta_z : X(\mathfrak{D}_{L|U_z}) \to X(\mathfrak{D}'_{L|V_z})$ . Indeed, identifying  $X(\mathfrak{D}_{L|U_z})$  with  $X(\mathfrak{D}'_{L|V_z}) \times_{V_z} U_z$  via étale base change, the morphism  $\delta_z$  is simply given by the first projection.

Let us denote by  $\Xi_L$  the colored cone ( $\mathscr{C}(\mathfrak{D}_L)_z, \mathcal{F}_L$ ) and by  $X_{\Xi_L}$  the corresponding embedding of  $\mathbb{G}_m \times L/H_L$ ; then  $X_{\Xi_L}$  is L-isomorphic to  $X(\mathfrak{D}'_L)$ . Let  $\gamma_z$  be the morphism which makes the diagram

$$\begin{array}{c|c} X(\mathfrak{D}_{L|U_z}) \xrightarrow{\gamma_z} & X_{\Xi_L} \\ & & & \downarrow \\ & & & \downarrow \\ & & & \downarrow \\ X(\mathfrak{D}'_{L|V_z}) \xrightarrow{\longleftarrow} & X(\mathfrak{D}'_L) \end{array}$$

commute, where the bottom arrow is an open embedding. In particular,  $\gamma_z$  is an étale morphism.

Let us prove  $(ii) \Rightarrow (i)$ . Denote by  $\Xi$  the colored cone  $(\mathscr{C}(\mathfrak{D})_z, \mathcal{F})$ , and let  $X_{\Xi}$  be the corresponding embedding of  $\mathbb{G}_m \times G/H$ . Then we have a G'-isomorphism  $X_{\Xi} \cong G' \times^{P'_{\mathcal{F}}} X_{\Xi_L}$  (see [Tim11, Th. 28.2]), where  $G' = \mathbb{G}_m \times G$  and  $P'_{\mathcal{F}} \subset G'$  is the parabolic subgroup constructed as in §2.1.1. By assumption,  $X_{\Xi}$  is smooth and thus so is  $X_{\Xi_L}$ . This implies that  $X(\mathfrak{D}_{L|U_z})$  is also smooth. Since  $C_0$  is affine,  $(X(\mathfrak{D}_{L|U_z}))_{z\in C_0}$  is an open covering of  $X(\mathfrak{D}_L)$ . Therefore  $X(\mathfrak{D}_L)$  is smooth and thus, by parabolic induction, so is  $X(\mathfrak{D})$ .

Let us prove  $(i) \Rightarrow (ii)$ . If  $X(\mathfrak{D})$  is smooth, then so is  $X(\mathfrak{D}_L)$  by parabolic induction. Hence, by the diagram above, there exists an open subset  $V \subset \mathbb{A}^1$ containing 0 such that  $X(\mathfrak{D}'_{L|V})$  is smooth. Denoting  $L' = \mathbb{G}_m \times L$  and identifying  $X(\mathfrak{D}'_{L|V})$  with an open subset of  $X_{\Xi_L}$ , we have  $L' \cdot X(\mathfrak{D}'_{L|V}) = X_{\Xi_L}$ . This implies that  $X_{\Xi_L}$  is smooth, and thus so is  $X_{\Xi}$ .

We say that two proper colored polyhedral divisors  $\mathfrak{D}$  and  $\mathfrak{D}'$  on  $C_0$  are equivalent if  $A[C_0, \mathfrak{D}]$  and  $A[C_0, \mathfrak{D}']$  (defined by (4)) are isomorphic as *M*-graded algebras; see [AH06, §8] and [Lan15, Prop. 4.5] for a combinatorial description of the equivalence between two such polyhedral divisors. If  $\mathfrak{D}$  and  $\mathfrak{D}'$  are equivalent this does not imply a priori that  $X(\mathfrak{D})$  and  $X(\mathfrak{D}')$  are isomorphic.

The next theorem gives a smoothness criterion for  $X(\mathfrak{D})$  when  $(\mathfrak{D}, \mathcal{F})$  is a proper colored polyhedral divisor on a projective curve  $C_0 = C$ .

**Theorem 2.6.** With the same notation as before, and assuming that  $C_0$  is projective (i.e.,  $C_0 = C$ ), the following statements are equivalent:

- (i) The G-variety  $X(\mathfrak{D})$  is smooth.
- (ii) The curve C is  $\mathbb{P}^1$ , the polyhedral divisor  $\mathfrak{D}$  is equivalent to a proper colored polyhedral divisor  $\mathfrak{D}^{0,\infty} = \sum_{z \in \mathbb{P}^1} \Delta_z \cdot [z]$  with  $\Delta_z = \sigma$  except when z = 0 or  $\infty$ , and the simple embedding of the  $\mathbb{G}_m \times G$ -homogeneous space  $\mathbb{G}_m \times G/H$ associated with the colored cones  $(\mathcal{C}, \mathcal{F})$  (see §1.3.1) is smooth, where  $\mathcal{C}$  is the cone generated by  $(\sigma \times \{0\}) \cup (\Delta_0 \times \{1\}) \cup (\Delta_\infty \times \{-1\})$ .

*Proof.* Let us prove  $(i) \Rightarrow (ii)$ . Suppose that  $X = X(\mathfrak{D})$  is smooth. Then by Lemma 2.1 we can assume that X is affine and identify k[X] with

$$\mathcal{A}[C,\mathfrak{D}] = \bigoplus_{m \in \sigma^{\vee} \cap M} H^0(C, \mathcal{O}_C(\mathfrak{D}(m))) \otimes_k V(m).$$

Moreover, we can assume that  $\sigma^{\vee}$  is strongly convex; otherwise, there is a nontrivial torus D and a G-variety X' such that  $X \cong D \times X'$  and then we replace X by X'. By Luna's slice theorem (see [Lun73, §III, Cor. 2]), there is a G-isomorphism  $X \cong G \times^F V$ , where  $F \subset G$  is a reductive closed subgroup and V is a F-module. It follows from the proof of Luna's slice theorem that F is in fact the stabilizer of a point of X and thus F is a horospherical subgroup. Since F is reductive and contains a maximal unipotent subgroup of G, it contains the semisimple part of G, and thus G/F is a torus. Now the surjective map  $G \times^F V \to G/F$  induces an inclusion  $k[G/F] \subset k[G \times^F V] = k[X]$ . Since  $\sigma^{\vee}$  is strongly convex and  $\mathfrak{D}$  is proper, we have k[G/F] = k and so G = F. Therefore we obtain that X is G-isomorphic to the G-module V. Let us identify X with V and let us denote by  $\gamma : G \to GL(X)$  the corresponding homomorphism. If  $T \subset B$  is a maximal torus, then there exists a maximal torus  $\mathbb{T} \subset GL(X)$  containing  $\gamma(T)$  and normalizing  $\gamma(U)$ , where U is the unipotent radical of B. Therefore X//U is a toric variety for the action of  $\mathbb{T}$ .

It follows from [AH06, §11] that  $C = \mathbb{P}^1$  and  $\mathfrak{D}$  is equivalent to a polyhedral divisor  $\mathfrak{D}^{0,\infty}$  supported by 0 and  $\infty$ . We fix an isomorphism of *M*-graded algebras  $\phi: A[C,\mathfrak{D}] \to A[C,\mathfrak{D}^{0,\infty}]$ . Then there exists a group homomorphism

$$(M, +) \rightarrow (k(C)^*, \times), \quad m \mapsto f_m$$

and a linear automorphism  $\delta: M \to M$  preserving  $\sigma^{\vee} \cap M$  such that for every  $m \in \sigma^{\vee} \cap M$ , we have  $\phi(f\chi^m) = ff_m^{-1}\chi^{\delta(m)}$ . Let us denote by the same letter  $\delta: N \to N$  the dual linear map of  $\delta$ . We define a new  $\sigma$ -polyhedral divisor on C by

$$\delta_{\star}(\mathfrak{D}^{0,\infty}) = \sum_{y \in C} \delta(\Delta_y^{0,\infty}) \cdot [y],$$

where  $\Delta_y^{0,\infty}$  is the coefficient of  $\mathfrak{D}^{0,\infty}$  at the point  $y \in C$ . Then  $\delta_*(\mathfrak{D}^{0,\infty})(m) = \mathfrak{D}^{0,\infty}(\delta(m))$  for every  $m \in \sigma^{\vee} \cap M$ , and we have a *G*-isomorphism

$$\mathcal{A}[C,\mathfrak{D}] \to \mathcal{A}[C,\delta_{\star}(\mathfrak{D}^{0,\infty})], \ f \otimes v \mapsto ff_m^{-1} \otimes v$$

with  $f \in k(C)^*$  and  $v \in V(m)$ . Therefore  $X = \operatorname{Spec} \mathcal{A}[C, \mathfrak{D}]$  identifies with the horospherical variety associated with the colored cone  $((\delta \times Id)(\mathcal{C}), \mathcal{F})$ . Hence, by the smoothness criterion of [BM13, §5], this is equivalent to the smoothness of the horospherical variety associated with the colored cone  $(\mathcal{C}, \mathcal{F})$ .

The converse implication  $(ii) \Rightarrow (i)$  is proved by using the same kind of arguments and is left to the reader.

The smoothness criteria of Theorems 2.5 and 2.6 can be made explicit by applying the smoothness criterion in the horospherical embedding case given by the following theorem; see [Pau83,  $\S3.5$ ], [Pas06,  $\S$ II], and [BM13,  $\S5$ ].

**Theorem 2.7.** Let X be a G-equivariant embedding of the horospherical homogeneous space G/H associated with a colored cone  $(\mathcal{C}, \mathcal{F})$  (see §1.3.1). Then X is smooth if and only if the following conditions are satisfied.

- (i) The elements of F have pairwise distinct images through the map ρ defined in §1.2.2.
- (ii) The cone C is generated by a subset of a basis of N containing  $\varrho(\mathcal{F})$ .
- (iii) We have the equality

$$|W_I| \cdot \prod_{\alpha \in I_{\mathcal{F}}} a_\alpha = |W_{I \cup I_{\mathcal{F}}}|,$$

where I is the subset of simple roots of G defined in §1.1,  $I_{\mathcal{F}} = \{\alpha \in S \setminus I \mid D_{\alpha} \in \mathcal{F}\}$ ,  $W_I$  is the subgroup of the Weyl group  $W = N_G(T)/T$  generated by the simple reflexions  $s_{\alpha}$  ( $\alpha \in I$ ),  $a_{\alpha} = \langle \sum_{\beta \in \Phi^+ \setminus \Phi_I} \beta, \hat{\alpha} \rangle$ ,  $\Phi^+$  is the set of positive roots with respect to (T, B), and  $\Phi_I$  is the set of roots that are sums of elements of I.

2.2. **Decoloration morphism.** In this section, we prove the existence of the decoloration morphism for the normal horospherical *G*-varieties of complexity one; see [Bri91, §3.3] for the spherical case. This will be used several times in a crucial way to prove our statements in the following, and also to construct an explicit resolution of singularities of  $X(\Sigma)$  in Proposition 2.21.

**Definition 2.8.** Let  $\Sigma = \{(\mathfrak{D}^i, \mathcal{F}^i)\}_{i \in J}$  be a colored divisorial fan, then the *decoloration* of  $\Sigma$  is the colored divisorial fan

$$\Sigma_{dec} \coloneqq \{ (\mathfrak{D}^i, \emptyset) \}_{i \in J}.$$

A *G*-model  $X(\Sigma)$  of *Z* such that  $\Sigma = \Sigma_{dec}$  is called *quasi-toroidal*. As a consequence of the description of the germs of a *G*-model of *Z*, this notion does not depend on the choice of  $\Sigma'$ . Likewise, if  $(\mathcal{C}_0, \mathcal{F}_0)$  is the colored datum (see §1.3.4) of some germ  $\Gamma' \subset X(\Sigma)$ , then we say that the germ  $\Gamma' \subset X(\Sigma_{dec})$  corresponding to the colored datum  $(\mathcal{C}_0, \emptyset)$  is the *decoloration* of  $\Gamma'$ .

Let  $\Sigma = \{(\mathfrak{D}^i, \mathcal{F}^i)\}_{i \in J}$  be a colored divisorial fan, then the collection of k-schemes  $Y(\mathfrak{D}^i) = \operatorname{Spec} A[C_0^i, \mathfrak{D}^i]$ , equipped with their K-action, glue together to give a normal K-variety of complexity one that we denote by  $Y(\Sigma)$ ; see [AHS08, Th. 5.3 and Remark 7.4 (ii)].

Before stating the next result, let us introduce some notation. Let  $\Gamma \subset X$  be a germ of a *G*-model of *Z*; we say that  $\Gamma$  is a *geometric realization* of  $\mathcal{O}_{X,\Gamma}$  in the variety *X*. The *support* of  $\Gamma$ , denoted by  $\operatorname{Supp}(\Gamma)$ , is the set of *G*-valuations *v* of k(Z) such that  $\mathcal{O}_v$  dominates the local ring  $\mathcal{O}_{X,\Gamma}$ .

**Proposition 2.9.** Let  $\Sigma = \{(\mathfrak{D}^i, \mathcal{F}^i)\}_{i \in J}$  be a colored divisorial fan, and let  $X_0^i$ and  $X_{dec}^i$  be the charts corresponding to  $(\mathfrak{D}^i, \mathcal{F}^i)$  and  $(\mathfrak{D}^i, \emptyset)$  respectively. The inclusions  $k[X_0^i] \subset k[X_{dec}^i]$  induce a proper birational *G*-morphism

$$\pi_{dec}: X(\Sigma_{dec}) \to X(\Sigma).$$

Moreover, for every germ  $\Gamma \subset X$ , the subset  $\pi_{dec}^{-1}(\Gamma)$  is the germ obtained by decoloring the colored datum of  $\Gamma$ . Also, there exists a G-isomorphism between  $X(\Sigma_{dec})$  and the twisted product  $G/H \times^K Y(\Sigma)$ , where  $Y(\Sigma)$  is the K-variety defined above and K = P/H acts on G/H as follows: for every  $g \in G$ , for every  $pH \in K$ , we have  $pH \cdot gH = gp^{-1}H$ .

*Proof.* Let us fix an index  $i \in J$ . The inclusion  $k[X_0^i] \subset k[X_{dec}^i]$  induces a birational *B*-morphism  $\iota_0 : X_{dec}^i \to X_0^i$ . The latter induces a *G*-equivariant birational map  $\iota : X(\mathfrak{D}^i, \emptyset) \to X(\mathfrak{D}^i, \mathcal{F}^i)$ . By [Tim11, Prop. 12.12], the morphism  $\iota$  extends to a *G*-morphism

$$\pi^i_{dec}: X(\mathfrak{D}^i, \emptyset) \to X(\mathfrak{D}^i, \mathcal{F}^i)$$

if and only if for every germ  $\Gamma \subset X(\mathfrak{D}^i, \emptyset)$ , there exists a (necessarily unique) germ  $\Gamma' \subset X(\mathfrak{D}^i, \mathcal{F}^i)$  such that the local ring  $\mathcal{O}_{X(\mathfrak{D}^i, \emptyset), \Gamma}$  dominates  $\mathcal{O}_{X(\mathfrak{D}^i, \mathcal{F}^i), \Gamma'}$ . Let us consider the colored datum  $(\mathcal{C}_0^i, \emptyset)$  of the germ  $\Gamma \subset X(\mathfrak{D}^i, \emptyset)$ , and denote

$$\mathcal{F}_0^i = \{ D \in \mathcal{F}^i \mid \varrho(D) \in \mathcal{C}_0^i \}$$

Then  $(\mathcal{C}_0^i, \mathcal{F}_0^i)$  is the colored datum of a germ  $\Gamma' \subset X(\mathfrak{D}^i, \mathcal{F}^i)$ . By [Tim11, Prop. 14.1 (2)] or [Kno93, §3.8], the support of a germ associated to a color datum  $(\mathcal{C}', \mathcal{F}')$  depends only on  $\mathcal{C}'$ , and thus  $\operatorname{Supp}(\Gamma) = \operatorname{Supp}(\Gamma')$ . This implies the equality  $\mathcal{O}_{X(\mathfrak{D}^i, \emptyset), \Gamma} = \mathcal{O}_{X(\mathfrak{D}^i, \mathcal{F}^i), \Gamma'}$ ; see the proof of [Tim11, Prop. 14.1(1)]. In particular,  $\mathcal{O}_{X(\mathfrak{D}^i, \emptyset), \Gamma}$  dominates  $\mathcal{O}_{X(\mathfrak{D}^i, \mathcal{F}^i), \Gamma'}$  and thus  $\iota$  extends to a G-morphism.

Let now  $\Gamma' \subset X(\mathfrak{D}^i, \mathcal{F}^i)$  be an arbitrary germ. Since the induced map  $(\pi^i_{dec})_*$ is the identity on  $\mathscr{E}$  (see the remark before [Tim11, Th. 12.13]), the subset  $(\pi^i_{dec})^{-1}(\Gamma')$  is irreducible and coincides with the decoloration of  $\Gamma'$ .

The properness of  $\pi^i_{dec}$  follows from the properness criterion given by [Tim11, Th. 12.13]. The existence and the properties of the morphism

$$\pi_{dec}: X(\Sigma_{dec}) \to X(\Sigma)$$

are then obtained by gluing.

For the last claim, it suffices to show that  $X(\mathfrak{D}, \emptyset)$  is *G*-isomorphic to  $G/H \times^K Y(\mathfrak{D})$ . The latter identifies with the *G*-variety  $G \times^P Y(\mathfrak{D})$  obtained by parabolic induction. Now it follows from [Tim11, Prop. 14.4 and 20.13] that  $X(\mathfrak{D}, \emptyset)$  and the  $G \times^P Y(\mathfrak{D})$  have the same combinatorics data, whence the result.  $\Box$ 

**Example 2.10.** We consider the natural action of  $G = SL_3$  on  $\mathbb{A}^3_* = \mathbb{A}^3 \setminus \{(0,0,0)\}$ . Let H be the isotropy subgroup of the point (1,0,0) for this action. Then H is a horospherical subgroup of G and  $\mathbb{A}^3_* \cong G/H$ . Also, the torus  $K = N_G(H)/H \cong \mathbb{G}_m$  acts diagonally on  $\mathbb{A}^3_*$  and the fibration  $G/H = \mathbb{A}^3_* \to G/P = \mathbb{P}^2$  is simply the quotient morphism for the  $\mathbb{G}_m$ -action.

Let us consider the colored  $\sigma$ -polyhedral divisor on  $\mathbb{A}^1 = \operatorname{Spec} k[t]$  defined by  $\mathcal{F} = \emptyset$  and  $\mathfrak{D} = [\frac{1}{2}, +\infty[\cdot[0]], \text{ where } N_{\mathbb{Q}} = \mathbb{Q}, \text{ and } \sigma = \mathbb{Q}_{\geq 0}.$  The k-algebra

$$A[\mathbb{A}^1,\mathfrak{D}] = \bigoplus_{m \ge 0} k[t]t^{-\lfloor \frac{1}{2}m \rfloor} \chi^m$$

is generated by the homogeneous elements  $t, \chi^1$ , and  $\frac{1}{t}\chi^2$ . Therefore,  $Y(\mathfrak{D}) =$ Spec  $A[\mathbb{A}^1, \mathfrak{D}]$  can be identified with the affine surface  $V(xz - y^2) \subset \mathbb{A}^3$  equipped with the  $\mathbb{G}_m$ -action defined by  $\lambda \cdot (x, y, z) = (x, \lambda^{-1}y, \lambda^{-2}z)$  with  $\lambda \in \mathbb{G}_m$ .

Denoting by  $(x_1, x_2, x_3)$  a system of coordinates of  $\mathbb{A}^3$ , the twisted action of  $\mathbb{G}_m$ on the product  $G/H \times Y(\mathfrak{D})$  is given by

$$\lambda \cdot (x_1, x_2, x_3, x, y, z) = (\lambda^{-1} x_1, \lambda^{-1} x_2, \lambda^{-1} x_3, x, \lambda^{-1} y, \lambda^{-2} z).$$

By Proposition 2.9, the *G*-variety  $X \coloneqq X(\mathfrak{D}, \emptyset)$  identifies with the quotient  $G/H \times^K Y(\mathfrak{D})$ . Hence, X is the hypersurface  $xy - z^2 = 0$  in the complement of

$$\{[0:0:0:x:y:z] | [x:y:z] \in \mathbb{P}(0,-1,-2)\}$$

in the weighted projective space  $X' := \mathbb{P}(-1, -1, -1, 0, -1, -2)$ . Let  $B \subset G$  be the Borel subgroup of upper triangular matrices. To determine a chart of X, it suffices to determine the inverse image of the open B-orbit in  $\mathbb{P}^2 = G/P$  through the projection  $q : X = G/H \times^K Y(\mathfrak{D}) \to G/P$ . The open orbit  $B\bar{x}_0 \subset \mathbb{P}^2$  is precisely  $\mathbb{P}^2 \setminus \{x_3 = 0\} \simeq \mathbb{A}^2$ . Thus the chart  $X_0(\mathfrak{D}) = q^{-1}(B\bar{x}_0)$  is the hypersurface  $xy - z^2 = 0$ in  $X' \setminus \{x_3 = 0\}$  which is isomorphic to  $\mathbb{A}^2 \times V(xy - z^2)$ .

2.3. Parametrization of the stable prime divisors. In this section, we start by describing in Theorem 2.11 the germs of codimension one of a normal horospherical G-variety of complexity one X. From this, we deduce a description of the class group of X by generators and relations; see Corollary 2.12. Next, we obtain a factoriality criterion for X; see Corollary 2.14. Finally, in §2.3.4, we relate the description of stable Cartier divisors obtained by Timashev in [Tim00] to our description of stable Weil divisors.

2.3.1. To state our results we need first to introduce the set of vertices and the set of extremal rays of a colored polyhedral divisor. Let  $(\mathfrak{D}, \mathcal{F})$  be an element of a colored divisorial fan  $\Sigma$  with  $\mathfrak{D} = \sum_{z \in C_0} \Delta_z \cdot [z]$ . The set of vertices of  $\mathfrak{D}$ , denoted by Vert $(\mathfrak{D})$ , consists in pairs (z, v) where  $z \in C_0$  and  $v \in \Delta_z(0)$  is a vertex of  $\Delta_z$ . If  $\Sigma = \{(\mathfrak{D}^i, \mathcal{F}^i)\}_{i \in J}$ , then we put

$$\operatorname{Vert}(\Sigma) \coloneqq \bigcup_{i \in J} \operatorname{Vert}(\mathfrak{D}^i) \subset C \times N_{\mathbb{Q}}.$$

The set of extremal rays of  $\mathfrak{D}$ , denoted by  $\operatorname{Ray}(\mathfrak{D})$  or  $\operatorname{Ray}(\mathfrak{D}, \mathcal{F})$ , consists in extremal rays  $\rho \subset \sigma$  such that  $\rho \cap \varrho(\mathcal{F}) = \emptyset$ , and satisfying  $\rho \cap \deg \mathfrak{D} = \emptyset$  when  $C_0 = C$ . To simplify the notation, we denote by the same letter an extremal ray of a polyhedral cone of  $N_{\mathbb{Q}}$  and its primitive vector with respect to the lattice N. We also denote

$$\operatorname{Ray}(\Sigma) \coloneqq \bigcup_{i \in J} \operatorname{Ray}(\mathfrak{D}^i) \subset N_{\mathbb{Q}},$$

where we recall that  $N_{\mathbb{Q}}$  naturally identifies with a subset of  $\mathscr{E}$ ; see §1.2.3. Finally, we denote by  $C_{\Sigma}$  the union of open subsets  $\bigcup_{i \in J} C_0^i \subset C$ , where  $C_0^i$  is the curve on which  $\mathfrak{D}^i \in \Sigma$  is defined.

2.3.2. In the next theorem we parametrize the set of *G*-divisors of a *G*-model  $X(\Sigma)$  of *Z* by the set  $Vert(\Sigma) \coprod Ray(\Sigma)$ . This description is a natural generalization of the case of normal *T*-varieties specialized to the case of *T*-actions of complexity one; see [FZ03, Th. 4.22] and [PS11, Prop. 3.13].

**Theorem 2.11.** Let  $Div(\Sigma)$  denote the set of G-divisors of  $X(\Sigma)$ . With the notation above, the map

$$\operatorname{Vert}(\Sigma) \coprod \operatorname{Ray}(\Sigma) \to \operatorname{Div}(\Sigma), \quad (z,v) \mapsto D_{(z,v)}, \ \rho \mapsto D_{\rho}$$

which to the vertex (z, v) associates the germ  $D_{(z,v)}$  of  $X(\Sigma)$  defined by the colored datum  $([(z, \mathbb{Q}_{\geq 0}(v, 1))], \emptyset)$  resp. to the ray  $\rho$  associates the germ  $D_{\rho}$  of  $X(\Sigma)$ defined by the colored datum  $(\rho, \emptyset) = ([(\cdot, \rho, 0)], \emptyset)$ , is a bijection (see §1.3.4 for the definition of colored datum). Proof. Without loss of generality, we can suppose that  $X(\Sigma) = X(\mathfrak{D})$  is given by a proper colored  $\sigma$ -polyhedral divisor  $(\mathfrak{D}, \mathcal{F})$  on a dense open subset  $C_0 \subset C$ , where  $\mathfrak{D} = \sum_{z \in C_0} \Delta_z \cdot [z]$ . By [Tim11, Th. 16.19 (2)], the maximal germs of  $X(\mathfrak{D})$  have color data of the form  $(\rho, \mathcal{F}_1)$  and  $([(z, \mathbb{Q}_{\geq 0}(v, 1))], \emptyset)$  where  $\rho \subset \sigma$  is an extremal ray,  $\mathcal{F}_1 = \{D_\alpha \in \mathcal{F} \mid \varrho(D_\alpha) \in \rho\}, z \in C_0$ , and  $v \in \Delta_z(0)$ . We are going to examine in each case the maximal germs corresponding to *G*-divisors. We proceed in three steps.

Step 1: We consider the case where  $X(\mathfrak{D})$  is quasi-toroidal, i.e.,  $\mathcal{F} = \emptyset$ . Then, by Proposition 2.9,  $X(\mathfrak{D})$  identifies with  $G/H \times^K Y(\mathfrak{D})$  as a *G*-variety. By [Jan87, I.5.21(1)], there is a natural bijection between the set of *K*-divisors on  $Y(\mathfrak{D})$  and the set of *G*-divisors on  $G/H \times^K Y(\mathfrak{D})$ . By [PS11, Prop. 3.13], the set of *K*-divisors on  $Y(\mathfrak{D})$  is parametrized by the set  $Vert(\mathfrak{D}) \coprod Ray(\mathfrak{D})$ :

$$(z,v) \mapsto \Gamma_{(z,v)}, \ \rho \mapsto \Gamma_{\rho}$$

The *G*-valuation  $v_{D_{\rho}}$  with  $D_{\rho} = G/H \times^{K} \Gamma_{\rho}$  resp.  $v_{D_{(z,v)}}$  with  $D_{(z,v)} = G/H \times^{K} \Gamma_{(z,v)}$ , is represented by  $[(\cdot, \rho, 0)] \in \mathscr{E}$  resp. by  $[(z, \mu(v)(v, 1))] \in \mathscr{E}$ , where  $\mu(v) := \inf\{d \in \mathbb{Z}_{>0} \mid dv \in \mathbb{N}\}$ . This parametrization of the *G*-divisors on  $X(\mathfrak{D})$  by the set  $\operatorname{Vert}(\mathfrak{D}) \coprod \operatorname{Ray}(\mathfrak{D})$  is the one requested.

Step 2: We now consider an arbitrary simple G-model  $X(\mathfrak{D}, \mathcal{F})$  of Z, and let

$$\pi_{dec}: X(\mathfrak{D}, \emptyset) \to X(\mathfrak{D}, \mathcal{F})$$

be the decoloration morphism. By Proposition 2.9, any *G*-divisor on  $X(\mathfrak{D}, \mathcal{F})$  is the image of a (unique) *G*-divisor on  $X(\mathfrak{D}, \emptyset)$ . Let us consider the *G*-divisor  $D'_{\rho}$ on  $X(\mathfrak{D}, \emptyset)$  corresponding to the color datum ([ $(\cdot, \rho, 0)$ ],  $\emptyset$ ), where  $\rho$  belongs to Ray( $\mathfrak{D}, \emptyset$ ). Then  $D_{\rho} = \pi_{dec}(D'_{\rho})$  is the germ of  $X(\mathfrak{D}, \mathcal{F})$  corresponding to the color datum ([ $(\cdot, \rho, 0)$ ],  $\mathcal{F}_1$ ). In this second step, we want to prove that  $D'_{\rho}$  is contracted by  $\pi_{dec}$  if and only if  $\mathcal{F}_1 \neq \emptyset$ .

Let  $\rho^* = \rho^{\perp} \cap \sigma^{\vee} \subset \sigma^{\vee}$  denote the dual face of  $\rho$ . By properness of  $\mathfrak{D}$  (see [AH06, §2]) there exists a homogeneous element  $f\chi^m \in A[C_0, \mathfrak{D}]$  of degree m belonging to the relative interior of  $\rho^*$  such that

$$\{z \in C_0 \mid \Delta_z \neq \sigma\} \subset Z(f) = \operatorname{Supp}(\operatorname{div} f + \mathfrak{D}(m)) \subset C_0.$$

The localization  $\mathfrak{D}_{f}^{m}$  of  $\mathfrak{D}$  with respect to  $f\chi^{m}$  is the colored  $\rho$ -polyhedral divisor trivial on the curve  $(C_{0})_{f}^{m} = C_{0} \setminus Z(f)$  with set of colors  $\mathcal{F}_{1}$ ; see Definition 1.5. We denote by X' the simple embedding of G/H associated with the colored cone  $(\rho, \mathcal{F}_{1})$  (see §1.3.1). By computing in an appropriate chart, one checks that the product  $(C_{0})_{f}^{m} \times X'$  identifies with  $X(\mathfrak{D}_{f}^{m}, \mathcal{F}_{1})$  as a G-variety. We denote by Gxthe (unique) closed orbit of X'. Then  $D_{\rho}'' \coloneqq (C_{0})_{f}^{m} \times Gx$  is a maximal germ (for the inclusion) of  $X(\mathfrak{D}_{f}^{m}, \mathcal{F}_{1})$  and corresponds thus to the colored datum  $(\rho, \mathcal{F}_{1})$ ; see [Tim11, Th. 16.19] for details. The germ  $D_{\rho}''$  is a geometric realization of  $\mathcal{O}_{X(\mathfrak{D},\mathcal{F}),D_{\rho}}$  in the open subset  $X(\mathfrak{D}_{f}^{m},\mathcal{F}_{1}) \subset X(\mathfrak{D},\mathcal{F})$ , that is,  $D_{\rho}$  is the closure of  $D_{\rho}''$  in  $X(\mathfrak{D},\mathcal{F})$ . Therefore,  $D_{\rho}$  is a divisor on  $X(\mathfrak{D},\mathcal{F})$  if and only if Gx is a divisor on X' if and only if  $\mathcal{F}_{1} = \emptyset$ ; see [Kno91, Lemma 2.4] for the last equivalence. Step 3: It remains to study the germs of the form  $D_{(z,v)} \not\subseteq X(\mathfrak{D}, \mathcal{F})$ . Let us fix a vertex  $(z, v) \in \operatorname{Vert}(\mathfrak{D})$ , and let us write  $\sigma^{\vee}$  as a union

$$\sigma^{\vee} = \bigcup_{w \in \Delta_z(0)} \sigma_w^{\vee}, \text{ where } \sigma_w^{\vee} = \left\{ m \in \sigma^{\vee} \mid \langle m, w \rangle = \min_{v' \in \Delta_z(0)} \langle m, v' \rangle \right\}.$$

The cones  $\sigma_w^{\vee}$  are pairwise distinct, generate a quasi-fan of  $\sigma^{\vee}$  (see [AH06, §1]) and are all of full dimension in  $M_{\mathbb{Q}}$ . Let  $m \in \sigma_v^{\vee} \cap M$  be a lattice vector in the relative interior of  $\sigma_v^{\vee}$ . By properness of  $\mathfrak{D}$ , and up to a change of m by a strictly positive integer multiple, we can choose a homogeneous element  $f\chi^m \in A[C_0, \mathfrak{D}]$  of degree m such that  $(C_0)_f^m = C_0 \setminus Z(f)$  contains z. Then  $\mathfrak{D}_f^m$  is of the form

$$\sum_{z'\in (C_0)_f^m} Q_{z'}\cdot [z'],$$

where  $Q_{z'} \subset N_{\mathbb{Q}}$  is a polytope for every  $z' \in (C_0)_f^m$  and  $Q_z = \{v\}$ . The set of colors of  $\mathfrak{D}_f^m$  is empty. By Step 1 applied to the open subset  $X(\mathfrak{D}_f^m, \emptyset) \subset X(\mathfrak{D}, \mathcal{F})$ , the germ  $D'_{(z,v)} \not\subseteq X(\mathfrak{D}_f^m, \emptyset)$  corresponding to the colored datum  $([(z, \mathbb{Q}_{\geq 0}(v, 1))], \emptyset))$ is of codimension one. Thus  $D_{(z,v)}$  is of codimension one as the closure of  $D'_{(z,v)}$  in  $X(\mathfrak{D}, \mathcal{F})$ . This proves the existence of the parametrization of the *G*-divisors.  $\Box$ 

**Corollary 2.12.** The class group  $Cl(X(\Sigma))$  is isomorphic to the abelian group

$$\operatorname{Cl}(C_{\Sigma}) \oplus \bigoplus_{(z,v) \in \operatorname{Vert}(\Sigma)} \mathbb{Z}D_{(z,v)} \oplus \bigoplus_{\rho \in \operatorname{Ray}(\Sigma)} \mathbb{Z}D_{\rho} \oplus \bigoplus_{\alpha \in S \smallsetminus I} \mathbb{Z}D_{\alpha},$$

where  $D_{\alpha} \subset X(\Sigma)$  is the color associated with  $\alpha \in S \setminus I$ , modulo the relations:

$$[z] = \sum_{(z,v)\in\operatorname{Vert}(\Sigma)} \mu(v) D_{(z,v)} \text{ and}$$
$$\sum_{(z,v)\in\operatorname{Vert}(\Sigma)} \mu(v) \langle m, v \rangle D_{(z,v)} + \sum_{\rho\in\operatorname{Ray}(\Sigma)} \langle m, \rho \rangle D_{\rho} + \sum_{\alpha\in S \smallsetminus I} \langle m, \varrho(D_{\alpha}) \rangle D_{\alpha} = 0,$$

where  $m \in M$ ,  $z \in C_{\Sigma}$ , and  $\mu(v) = \inf\{d \in \mathbb{Z}_{>0} \mid dv \in \mathbb{N}\}.$ 

*Proof.* By [Tim11, Prop. 17.1], every divisor  $X(\Sigma)$  is linearly equivalent to a *B*-stable divisor. Hence, by Theorem 2.11, we have a surjective homomorphism  $\pi_{\Sigma}$  from the free abelian group

(5) 
$$\Phi_{\Sigma} \coloneqq \bigoplus_{(z,v) \in \operatorname{Vert}(\Sigma)} \mathbb{Z}D_{(z,v)} \oplus \bigoplus_{\rho \in \operatorname{Ray}(\Sigma)} \mathbb{Z}D_{\rho} \oplus \bigoplus_{\alpha \in S \smallsetminus I} \mathbb{Z}D_{\alpha}$$

onto  $\operatorname{Cl}(X(\Sigma))$ . The kernel of  $\pi_{\Sigma}$  is formed by principal divisors associated with the *B*-eigenvectors of  $k(X(\Sigma))$ , i.e., by elements of the form  $\operatorname{div}(f\chi^m)$ :

$$\sum_{(z,v)\in\operatorname{Vert}(\Sigma)} v_{D_{(z,v)}}(f\chi^m) \cdot D_{(z,v)} + \sum_{\rho\in\operatorname{Ray}(\Sigma)} v_{D_\rho}(f\chi^m) \cdot D_\rho + \sum_{\alpha\in S\smallsetminus I} v_{D_\alpha}(f\chi^m) \cdot D_\alpha$$
$$= \sum_{(z,v)\in\operatorname{Vert}(\Sigma)} \mu(v)(\langle m,v\rangle + \operatorname{ord}_z f)D_{(z,v)} + \sum_{\rho\in\operatorname{Ray}(\Sigma)} \langle m,\rho\rangle D_\rho + \sum_{\alpha\in S\smallsetminus I} \langle m,\varrho(D_\alpha)\rangle D_\alpha,$$

where  $f \in k(C)^*$  and  $m \in M$ . Let us now consider the surjective homomorphism from

$$\operatorname{Cl}(C_{\Sigma}) \oplus \bigoplus_{(z,v) \in \operatorname{Vert}(\Sigma)} \mathbb{Z} D_{(z,v)} \oplus \bigoplus_{\rho \in \operatorname{Ray}(\Sigma)} \mathbb{Z} D_{\rho} \oplus \bigoplus_{\alpha \in S \smallsetminus I} \mathbb{Z} D_{\alpha}$$

20

onto  $\Phi_{\Sigma}$  / Ker  $\pi_{\Sigma}$  defined by

$$\sum_{z \in C_{\Sigma}} a_z \cdot [z] + \sum_i a_i D_i \mapsto \sum_{z \in C_{\Sigma}} a_z \left( \sum_{(z,v) \in \operatorname{Vert}(\Sigma)} \mu(v) [D_{(z,v)}] \right) + \sum_i a_i [D_i],$$

where the  $D_i$  represent the *B*-divisors on  $X(\Sigma)$ . Then one may check that the kernel of this homomorphism is exactly given by the relations stated above.  $\Box$ 

**Example 2.13.** Returning to the example of the  $SL_3$ -variety

$$X(\mathfrak{D}) = \{xz - y^2 = 0\} \cap (\mathbb{P}(-1, -1, -1, 0, -1, -2) \setminus \mathbb{P}(0, -1, -2))$$

considered in §2.2, we can apply Corollary 2.12 to determine the class group of  $X(\mathfrak{D})$ . We obtain that  $\operatorname{Cl}(X(\mathfrak{D}))$  is the abelian group  $\mathbb{Z}D_{(0,\frac{1}{2})} \oplus \mathbb{Z}D_{\rho} \oplus \mathbb{Z}D_{\alpha}$ , where  $\rho = \mathbb{Q}_{\geq 0}$  and  $D_{\alpha}$  is the unique color of  $X(\mathfrak{D})$ , modulo the following relations:

- $2D_{(0,\frac{1}{2})} = 0$ ; and
- $mD_{\rho} + 2mD_{\alpha} = 0$  for every  $m \in \mathbb{Z}$ .

It follows that  $\operatorname{Cl}(X(\mathfrak{D})) \cong \mathbb{Z} \oplus \mathbb{Z}/2\mathbb{Z}$ .

2.3.3. Let  $Y(\Sigma)$  be the *K*-variety defined at the beginning of §2.2. We denote by  $\rho \mapsto \Gamma_{\rho}$ ,  $(z, v) \mapsto \Gamma_{(z,v)}$  the parametrization of the *K*-divisors on  $Y(\Sigma)$  given by Theorem 2.11. We recall that a normal variety is called *factorial* if its class group is trivial. The next corollary gives a criterion of factoriality for the *G*-variety  $X(\mathfrak{D})$ .

**Corollary 2.14.** Let  $(\mathfrak{D}, \mathcal{F})$  be a proper colored  $\sigma$ -polyhedral divisor on a dense open subset  $C_0 \subset C$ . Then  $X(\mathfrak{D})$  is factorial if and only if the two following conditions are satisfied.

- (i) The equality  $\operatorname{Cl}(Y(\mathfrak{D})) = \sum_{\Gamma_{\rho} \in \Gamma} \mathbb{Z}[\Gamma_{\rho}]$  holds, where  $\Gamma$  denotes the union of the K-divisors  $\Gamma_{\rho}$  with  $\rho$  satisfying  $\varrho(\mathcal{F}) \cap \rho \neq \emptyset$ .
- (ii) For every  $\alpha \in S \setminus I$ , there exists  $m_{\alpha} \in M$  and  $f_{\alpha} \in k(C)^{\star}$  such that

 $\langle m_{\alpha}, \varrho(D_{\alpha}) \rangle = 1,$ 

and for all  $\beta \in S \setminus (I \cup \{\alpha\}), (z, v) \in Vert(\mathfrak{D}), and \rho \in Ray(\mathfrak{D}, \mathcal{F}):$ 

$$\langle m_{\alpha}, \varrho(D_{\beta}) \rangle = \mu(v)(\langle m_{\alpha}, v \rangle + \operatorname{ord}_{z}(f_{\alpha})) = \langle m_{\alpha}, \rho \rangle = 0.$$

*Proof.* The decoloration morphism induces an isomorphism of varieties

$$X(\mathfrak{D},\mathcal{F})\backslash\Gamma_0\simeq X(\mathfrak{D},\varnothing)\backslash\pi_{dec}^{-1}(\Gamma_0),$$

where  $\Gamma_0$  is the *G*-stable closed subset

$$\left\{x \in X(\mathfrak{D}, \mathcal{F}) \mid Gx \subset \bigcup_{\alpha \in S \setminus I} D_{\alpha}\right\}.$$

As the codimension of  $\Gamma_0$  in  $X(\mathfrak{D}, \mathcal{F})$  is at least two, if  $\Gamma' \subset X(\mathfrak{D}, \emptyset)$  is the union of irreducible components of codimension one of  $\pi_{dec}^{-1}(\Gamma_0)$ , then we have a group isomorphism

$$\operatorname{Cl}(X(\mathfrak{D},\mathcal{F})) \simeq \operatorname{Cl}(X(\mathfrak{D},\varnothing) \backslash \Gamma').$$

By Proposition 2.9, we can identify  $X(\mathfrak{D}, \emptyset)$  with  $G/H \times^K Y(\mathfrak{D})$  and  $\Gamma'$  with  $G/H \times^K \Gamma$ . Let us consider the chart  $X_0 = q^{-1}(B\bar{x}_0)$ , where  $B\bar{x}_0$  is the open *B*-orbit of G/P and

$$q: G/H \times^K Y(\mathfrak{D}) = G \times^P Y(\mathfrak{D}) \to G/P$$

is the projection. The complement of the union of colors  $\bigcup_{\alpha \in S \setminus I} D_{\alpha}$  in  $X(\mathfrak{D}, \emptyset) \setminus \Gamma'$ is exactly

$$X_0 \setminus (X_0 \cap \Gamma') \simeq B\bar{x}_0 \times (Y(\mathfrak{D}) \setminus \Gamma).$$

As  $B\bar{x}_0$  is an affine space, by reiterating several times [Har77, Prop. II.6.6], we obtain

$$\operatorname{Cl}(X_0 \backslash \Gamma') \simeq \operatorname{Cl}(Y(\mathfrak{D}) \backslash \Gamma)$$

To sum up, we have an exact sequence:

$$\bigoplus_{\alpha \in S \setminus I} \mathbb{Z}D_{\alpha} \to \mathrm{Cl}\left(X(\mathfrak{D}, \emptyset) \backslash \Gamma'\right) \to \mathrm{Cl}\left(Y(\mathfrak{D}) \backslash \Gamma\right) \to 0,$$

where the first arrow is induced by the surjective homomorphism from the group  $\Phi_{\Sigma}$  defined by (5) onto  $\operatorname{Cl}(X(\mathfrak{D}))$ . Therefore  $\operatorname{Cl}(X(\mathfrak{D})) = 0$  if and only if  $\operatorname{Cl}(Y(\mathfrak{D}) \setminus \Gamma) = 0$  and the first arrow above is zero. By Corollary 2.12, this corresponds exactly to the conditions (*i*) and (*ii*). This proves the Corollary.

Remark 2.15. In general, the factoriality of  $Y(\mathfrak{D})$  does not imply the one of  $G \times^P Y(\mathfrak{D})$ . For instance, we can consider  $G = \mathrm{SL}_2$ , H the unipotent subgroup of upper triangular matrices,  $C = \mathbb{P}^1$ , and  $(\mathfrak{D}, \emptyset)$  the colored  $\mathbb{Q}_{\geq 0}$ -polyhedral divisor trivial on  $\mathbb{A}^1 \subset \mathbb{P}^1$ . The SL<sub>2</sub>-variety  $X(\mathfrak{D})$  identifies with  $\mathbb{A}^1 \times \mathrm{Bl}_0(\mathbb{A}^2)$ , where  $\mathrm{Bl}_0(\mathbb{A}^2)$  is the bowing-up of  $\mathbb{A}^2$  at the origin. We have  $\mathrm{Cl}(X(\mathfrak{D})) \simeq \mathbb{Z}$  whereas  $\mathrm{Cl}(Y(\mathfrak{D})) = \mathrm{Cl}(\mathbb{A}^2) = 0$ .

2.3.4. As a by-product of Theorem 2.11, we can refine the description of B-stable Cartier divisors of [Tim00] to our setting. Let us start with a definition.

**Definition 2.16.** Let  $\Sigma = \{(\mathfrak{D}^i, \mathcal{F}^i)\}_{i \in J}$  be a colored divisorial fan on C, and let  $\mathfrak{D} = \mathfrak{D}^i \in \Sigma$  be a colored polyhedral divisor on  $C_0$  with set of colors  $\mathcal{F} = \mathcal{F}^i$ . Recall that we denote by  $\mathscr{C}(\mathfrak{D})$  the hypercone associated with  $\mathfrak{D}$  (see Remark 1.7). An *integral linear function* on  $\mathfrak{D}$  is a map

$$\theta:\mathscr{C}(\mathfrak{D})\to\mathbb{Q}$$

satisfying the following properties:

(i) For every  $z \in C_0$ , there exists  $m_z \in M$  and  $\gamma_z \in \mathbb{Z}$  such that  $\theta(z, u, l) = u(m_z) + l\gamma_z$ , for every  $(u, l) \in \mathscr{C}(\mathfrak{D})_z$ .

If  $C_0 = C$ , then  $\theta$  must satisfy the extra condition:

(ii) We have  $m \coloneqq m_z = m_{z'}$ , for every  $z, z' \in C$ , and there exists  $f \in k(C)^*$  such that

$$\operatorname{div} f = \sum_{z \in C} \gamma_z \cdot [z].$$

Let us denote by  $\mathcal{F}_{\Sigma}$  the union of all the sets  $\mathcal{F}^{i}$ , where the  $(\mathfrak{D}^{i}, \mathcal{F}^{i})$  run over  $\Sigma$ . A colored integral piecewise linear function on  $\Sigma$  is a pair  $(\theta, (r_{\alpha}))$ , where  $\theta$  is a function

$$\theta: |\Sigma| \left(= \bigcup_{i \in J} \mathscr{C}(\mathfrak{D}^i)\right) \to \mathbb{Q}$$

such that the restriction  $\theta_{|\mathscr{C}(\mathfrak{D}^i) \cap \mathscr{C}(\mathfrak{D}^j)}$  is integral linear for every  $i, j \in J$ , and where  $(r_\alpha)$  is a sequence of integers with  $\alpha$  running over the set of roots  $\alpha \in S \setminus I$  such that  $D_\alpha \notin \mathcal{F}_{\Sigma}$ .

The pair  $(\theta, (r_{\alpha}))$  is called *principal* if  $\theta$  satisfies (ii) and  $r_{\alpha} = \langle m, \varrho(D_{\alpha}) \rangle$ . We denote respectively by  $PL(\Sigma)$  and  $Prin(\Sigma)$  the abelian groups (for the natural additive law) of colored integral piecewise linear functions of  $\Sigma$  and of principal colored integral piecewise linear functions of  $\Sigma$ . If  $\Sigma$  has a single element  $\mathfrak{D}$ , then we will denote  $PL(\mathfrak{D})$  and  $Prin(\mathfrak{D})$  instead of  $PL(\Sigma)$  and  $Prin(\Sigma)$  respectively.

As a direct consequence of [Kno94], [Tim00, §4], [Tim11, §17], and Theorem 2.11, we obtain the next result. See [Bri89, §3.1] for the spherical case and [PS11, Cor. 3.19] for the case of normal T-varieties.

**Corollary 2.17.** With the notation above, if  $(\theta, (r_{\alpha})) \in PL(\Sigma)$ , then

$$D_{\theta} \coloneqq \sum_{(z,v) \in \operatorname{Vert}(\Sigma)} \theta(z, \mu(v)(v, 1)) \cdot D_{(z,v)} + \sum_{\rho \in \operatorname{Ray}(\Sigma)} \theta(\cdot, \rho, 0) \cdot D_{\rho} + \sum_{D_{\alpha} \notin \mathcal{F}_{\Sigma}} \theta(\cdot, \varrho(D_{\alpha}), 0) \cdot D_{\alpha} + \sum_{D_{\alpha} \notin \mathcal{F}_{\Sigma}} r_{\alpha} \cdot D_{\alpha}$$

is a B-stable Cartier divisor on  $X(\Sigma)$ . More precisely, the map  $\theta \mapsto D_{\theta}$  is an isomorphism between the group  $PL(\Sigma)$  and the group of B-stable Cartier divisors on  $X(\Sigma)$ , and there is a short exact sequence:

$$0 \to \operatorname{Prin}(\Sigma) \to \operatorname{PL}(\Sigma) \to \operatorname{Pic}(X(\Sigma)) \to 0,$$

where  $\operatorname{Pic}(X(\Sigma))$  is the Picard group of  $X(\Sigma)$ .

2.4. Canonical class and log-terminal singularities. In this section, we give an explicit representative of the canonical class for X a normal horospherical Gvariety of complexity one; see Theorem 2.18. From this, we deduce a criterion for Xto be Q-Gorenstein; see Corollary 2.19. Then we construct an explicit resolution of singularities of X; see Proposition 2.21. Finally, we obtain a criterion to determine whether the singularities of X are log-terminal; see Theorem 2.22.

2.4.1. The next result gives an explicit canonical divisor for any normal horospherical *G*-variety of complexity one; see [Mos90, §5] for the case of the embeddings of  $SL_2$ , [Bri93] for the case of the spherical varieties, and [FZ03, Cor. 4.25], [PS11, Th. 3.21] for the case of normal *T*-varieties.

**Theorem 2.18.** Let  $\Sigma$  be a colored divisorial fan on C. Then with the notation of §2.3.1 every canonical divisor on  $X = X(\Sigma)$  is linearly equivalent to

$$K_X = -\sum_{\rho \in \operatorname{Ray}(\Sigma)} D_\rho + \sum_{(z,v) \in \operatorname{Vert}(\Sigma)} (\mu(v)b_z + \mu(v) - 1)D_{(z,v)} - \sum_{\alpha \in S \setminus I} a_\alpha D_\alpha,$$

where  $K_C = \sum_{z \in C} b_z \cdot [z]$  is a canonical divisor on C,  $a_\alpha = \langle \sum_{\beta \in \Phi^+ \setminus \Phi_I} \beta, \hat{\alpha} \rangle \ge 2$ ,  $\Phi^+$  is the set of positive roots with respect to (T, B), and  $\Phi_I$  is the set of roots that are sums of elements of I.

*Proof.* Let us consider the union

$$\Gamma' = \Gamma_0 \cup \operatorname{Sing}(X) \subset X,$$

where  $\Gamma_0$  is the biggest *G*-stable closed subset contained in the union of all the colors of *X* and Sing(*X*) is the singular locus. Then  $X(\Sigma') = X \setminus \Gamma'$  is smooth, quasi-toroidal and the set  $\operatorname{Vert}(\Sigma) \coprod \operatorname{Ray}(\Sigma)$  identifies with the set  $\operatorname{Vert}(\Sigma') \coprod \operatorname{Ray}(\Sigma')$ . Therefore, by Proposition 2.9, we can assume without loss of generality that  $X = G \times^P Y(\Sigma)$  is smooth and quasi-toroidal. Adapting the argument of the proof of [ST99, Prop. 4.2 c)] to our setting, we obtain an isomorphism of  $\mathcal{O}_X$ -modules

$$\mathcal{O}_X(K_X) \simeq \mathcal{O}_X(D) \otimes q^* \mathcal{O}_{G/P}(K_{G/P}),$$

where  $K_{G/P}$  is a canonical divisor on G/P given by

$$K_{G/P} = -\sum_{\alpha \in S \smallsetminus I} a_{\alpha} D_{\alpha}$$

and the divisor D is defined by

$$D = \sum_{(z,v)\in \operatorname{Vert}(\Sigma)} c_{(z,v)} D_{(z,v)} + \sum_{\rho \in \operatorname{Ray}(\Sigma)} c_{\rho} D_{\rho}$$

such that

$$K_{Y(\Sigma)} = \sum_{(z,v)\in\operatorname{Vert}(\Sigma)} c_{(z,v)}\Gamma_{(z,v)} + \sum_{\rho\in\operatorname{Ray}(\Sigma)} c_{\rho}\Gamma_{\rho}.$$

is a canonical divisor on  $Y(\Sigma)$  (see the beginning of §2.3.3 for the notation). We conclude by [PS11, Th. 3.21].

2.4.2. A normal *G*-variety *X* is called  $\mathbb{Q}$ -*Gorenstein* if one (and thus any) canonical divisor  $K_X$  is  $\mathbb{Q}$ -Cartier. The next corollary gives a combinatorial criterion for a normal horospherical *G*-variety of complexity one to be  $\mathbb{Q}$ -Gorenstein; see [Bri93, Prop. 4.1] for the spherical case and [LS13, Prop. 4.3] for the case of normal *T*-varieties.

**Corollary 2.19.** With the notation of §§2.3.1 and 2.3.4, the variety  $X(\Sigma)$  is  $\mathbb{Q}$ -Gorenstein if there exists  $d \in \mathbb{Z}_{>0}$  and  $\theta \in PL(\Sigma)$  such that the following conditions are all satisfied.

- (i) For every  $\rho \in \operatorname{Ray}(\Sigma)$ , we have  $\theta(\cdot, \rho, 0) = -d$ .
- (ii) There exists a canonical divisor  $K_C = \sum_{z \in C} b_z \cdot [z]$  on C such that, for every  $(z, v) \in \operatorname{Vert}(\Sigma)$ , we have  $\theta(z, \mu(v)v, 1) = d(\mu(v)b_z + \mu(v) 1)$ .
- (iii) For every  $D_{\alpha} \in \mathcal{F}_{\Sigma}$ , we have  $\theta(\cdot, \varrho(D_{\alpha}), 0) = -da_{\alpha}$ .

*Proof.* It is a straightforward consequence of Corollary 2.17 and Theorem 2.18.  $\Box$ 

2.4.3. The remainder of this paper is dedicated to the study of log-terminal singularities of a simple G-model of Z.

Let X be a Q-Gorenstein variety, let  $\phi: X' \to X$  be a resolution of singularities, and let  $d \in \mathbb{Z}_{>0}$  such that  $dK_X$  is Cartier. Then the pull-back  $\phi^*(dK_X)$  is welldefined. The *discrepancy* of  $\phi$  is the Q-divisor

$$K_{X'} - \phi^*(K_X) \coloneqq K_{X'} - \frac{1}{d}\phi^*(dK_X).$$

We say that X has (purely) log-terminal singularities if each coefficient of  $K_{X'} - \phi^*(K_X)$  is strictly bigger than -1. The property of having log-terminal singularities does not depend on the choice of the resolution of singularities  $\phi$ . More generally, if D is a Q-divisor on X such that  $K_X + D$  is Q-Cartier, then we say that the pair (X, D) is (purely) log-terminal if each coefficient of  $K_{X'} - \phi^*(K_X + D)$  is strictly bigger than -1.

The next lemma will be useful to prove Theorem 2.22; see [Kol97, §3] for more details about log-terminal singularities and for a proof of the statement.

**Lemma 2.20.** Let  $\phi : X' \to X$  be a proper birational morphism between normal varieties. Let D be a  $\mathbb{Q}$ -divisor on X such that  $K_X + D$  is  $\mathbb{Q}$ -Cartier, and let D' be the  $\mathbb{Q}$ -divisor on X' defined by

$$K_{X'} + D' = \phi^* (K_X + D).$$

Then (X, D) is log-terminal if and only if (X', D') is log-terminal and the coefficients of the prime divisors of -D' (corresponding to exceptional divisors of  $\phi$ ) are strictly bigger than -1.

2.4.4. In this subsection, we give an explicit method to construct a resolution of singularities of the *G*-model  $X(\mathfrak{D})$  of *Z*.

Let  $(\mathfrak{D}, \mathcal{F})$  be a proper colored  $\sigma$ -polyhedral divisor on a dense open subset  $C_0 \subset C$ . With the notation of §1.3, we denote

$$\tilde{Y}(\mathfrak{D}) = \operatorname{Spec}_{\mathcal{O}_C} \left( \bigoplus_{m \in \sigma^{\vee} \cap M} \mathcal{O}_C(\mathfrak{D}(m)) \chi^m \right).$$

Then the natural morphism  $\tilde{Y}(\mathfrak{D}) \to Y(\mathfrak{D})$  induces a partial desingularization

$$G \times^{P} \tilde{Y}(\mathfrak{D}) \to G \times^{P} Y(\mathfrak{D}) = X(\mathfrak{D}, \emptyset);$$

see [AH06, Th. 3.1 (ii)] and [LS13b, Construction 2.8]. The *G*-variety  $G \times^P \tilde{Y}(\mathfrak{D})$  identifies with  $X(\Sigma_{tor})$ , where  $\Sigma_{tor}$  is the colored divisorial fan  $\Sigma_{tor} = \{(\mathfrak{D}_{|C_i}, \emptyset)\}_{i \in J}$ ; the sequence  $(C_i)_{i \in J}$  forming a covering of  $C_0$  by affine open subsets. Moreover, if we consider a divisorial fan  $\overline{\Sigma}$  that refines  $\Sigma_{tor}$  and such that for all colored polyhedral divisor  $\mathfrak{D} \in \overline{\Sigma}$  and  $z \in C$  with  $\mathcal{C}(\mathfrak{D})_z \neq \emptyset$ , the polyhedral cone  $\mathcal{C}(\mathfrak{D})_z$  is regular (i.e., a cone generated by a subset of a basis of the lattice  $N \times \mathbb{Z}$ ), then  $X(\overline{\Sigma}) \to X(\Sigma_{tor})$  is a resolution of singularities. The next result is a straightforward consequence of this discussion.

**Proposition 2.21.** The morphism

$$\phi: X(\Sigma) \to X(\Sigma_{\mathrm{tor}}) \to X(\mathfrak{D}, \emptyset) \to X(\mathfrak{D}, \mathcal{F})$$

obtained by composing the decoloration morphism of §2.2 with the morphisms defined above is a resolution of singularities of  $X(\mathfrak{D}, \mathcal{F})$ . Moreover, with the notation of §2.3.1, the exceptional divisors of  $\phi$  correspond to the subsets  $\operatorname{Ray}(\overline{\Sigma})\setminus\operatorname{Ray}(\mathfrak{D})$ and  $\operatorname{Vert}(\overline{\Sigma})\setminus\operatorname{Vert}(\mathfrak{D})$ .

2.4.5. The next statement gives a characterization of the normal horospherical G-varieties of complexity one having log-terminal singularities. See [Bri93, Th. 4.1] for the spherical case and [LS13, Th. 4.9] for the case of normal T-varieties.

**Theorem 2.22.** Let  $(\mathfrak{D}, \mathcal{F})$  be a proper colored  $\sigma$ -polyhedral divisor on a dense open subset  $C_0 \subset C$ . We suppose that  $X(\mathfrak{D})$  is  $\mathbb{Q}$ -Gorenstein. Then  $X(\mathfrak{D})$  has log-terminal singularities if and only if one of the following assertions holds.

- (i) The curve  $C_0$  is affine.
- (ii) The curve  $C_0$  is the projective line  $\mathbb{P}^1$  and  $\sum_{z \in C_0} \left(1 \frac{1}{\mu_z}\right) < 2$ , where for every  $z \in C_0$  we denote  $\mu_z \coloneqq \max\{\mu(v) \mid v \in \Delta_z(0)\}$  and  $\mu(v) \coloneqq \inf\{d \in \mathbb{Z}_{>0} \mid dv \in \mathbb{N}\}.$

Proof. If  $C_0$  is affine, then by the proof of Theorem 2.5, we obtain that  $X(\mathfrak{D})$  is covered by étale open subsets of horospherical embeddings. Hence, by [Bri93, Th. 4.1],  $X(\mathfrak{D})$  has log terminal singularities. Therefore, we can assume that  $C_0 = C$  is projective. Let us consider a canonical divisor  $K_X$  as in Theorem 2.18, let  $d \in \mathbb{Z}_{>0}$ be such that  $dK_X$  is a Cartier divisor, and let  $\theta \in PL(\mathfrak{D})$  such that  $dK_X = D_{\theta}$ . By Corollary 2.17, we know that the restriction of  $dK_X$  on the open subset

$$X_1 \coloneqq X(\mathfrak{D}) \setminus \bigcup_{D_{\alpha} \notin \mathcal{F}_{\Sigma}} D_{\alpha}$$

is a principal divisor. Moreover, since every color  $D_{\alpha}$  satisfying  $D_{\alpha} \notin \mathcal{F}_{\Sigma}$  does not contain a *G*-orbit of  $X(\mathfrak{D})$ , the open subset  $\psi^{-1}(X_1)$  intersects each exceptional divisor of the partial desingularization  $\psi : X(\Sigma_{tor}) \to X(\mathfrak{D})$  given by Proposition 2.21. Therefore we can replace X by  $X_1$  and suppose that  $dK_X$  is principal. It follows that  $\psi^*(dK_X)$  is the principal divisor of a homogeneous element  $f\chi^m \in A_M$ of degree m considered as a rational function of  $X(\Sigma_{tor})$ . Hence, we have the equality

$$-D' \coloneqq K_{X(\Sigma_{\text{tor}})} - \psi^* K_{X(\mathfrak{D})} = \sum_{\rho \notin \text{Ray}(\mathfrak{D})} (-1 - \langle m, \rho \rangle) D_{\rho}.$$

By Lemma 2.20,  $X(\mathfrak{D})$  has log-terminal singularities if and only if -D' has its coefficients strictly bigger than -1 and  $(X(\Sigma_{tor}), D')$  has log-terminal singularities. Thus, by the same argument as in [LS13, §4] (see the sketch of proof before [LS13, Th. 4.9]), we know that  $X(\mathfrak{D})$  has log-terminal singularities if and only if  $\langle m, \rho \rangle < 0$ , for every  $\rho \notin \operatorname{Ray}(\mathfrak{D})$ .

Let  $\rho \subset \sigma$  be an extremal ray such that  $\rho \notin \operatorname{Ray}(\mathfrak{D})$ , then  $\rho \cap \varrho(\mathcal{F}) \neq \emptyset$  or  $\rho \cap \deg \mathfrak{D} \neq \emptyset$ . Let us suppose that  $\rho \cap \varrho(\mathcal{F}) \neq \emptyset$ . Then there exists  $D_{\alpha} \in \mathcal{F}$  and  $\lambda \in \mathbb{Q}_{>0}$  such that  $\rho = \lambda \varrho(D_{\alpha})$ . Hence

$$\langle m, \rho \rangle = \lambda \langle m, \rho(D_{\alpha}) \rangle = -\lambda da_{\alpha} < 0.$$

Let us now suppose that  $\rho \cap \deg \mathfrak{D} \neq \emptyset$ . Then  $\rho = \lambda v$  for a vertex  $v \in \deg \mathfrak{D}$ and for some  $\lambda \in \mathbb{Q}_{>0}$ . Let  $(v_z)_{z \in C}$  be a sequence of elements of  $\Delta_z(0)$  such that  $v = \sum_{z \in C} v_z$ . As the coefficients of  $K_X$  and  $\frac{1}{d} \operatorname{div} f \chi^m$  at the prime divisor  $D_{(z,v_z)}$  corresponding to any  $(z, v_z) \in \operatorname{Vert}(\mathfrak{D})$  are the same, we have equalities

$$\mu(v_z)b_z + \mu(v_z) - 1 = \frac{1}{d}\mu(v_z)(\langle m, v_z \rangle + \operatorname{ord}_z f),$$

where  $K_C = \sum_{z \in C} b_z \cdot [z]$  is a canonical divisor on C. Since C is projective, we have deg (div f) = 0. Hence, summing over C on both sides gives the equality

$$\deg K_C + \sum_{z \in C} \left( 1 - \frac{1}{\mu(v_z)} \right) = \frac{1}{d} \langle m, v \rangle.$$

As deg  $\mathfrak{D} \subset \sigma$ , we conclude that  $X(\mathfrak{D})$  has log-terminal singularities if and only if the condition (*ii*) is satisfied.

#### References

- [AH06] K. Altmann, J. Hausen. Polyhedral divisors and algebraic torus actions. Math. Ann. 334 (2006), no. 3, 557–607.
- [AHS08] K. Altmann, J. Hausen, H. Suess. Gluing affine torus actions via divisorial fans. Transform. Groups 13 (2008), no. 2, 215–242.
- [AIPSV12] K. Altmann, N. Ilten, L. Petersen, H. Suess, R. Vollmert. The geometry of T-varieties. Contributions to algebraic geometry, 17–69, EMS Ser. Congr. Rep., Eur. Math. Soc., Zürich, 2012.
- [Arz97] I. Arzhantsev. Actions of the group SL<sub>2</sub> that are of complexity one. (Russian) Izv. Ross. Akad. Nauk Ser. Mat. 61 (1997), no. 4, 3–18; translation in Izv. Math. 61 (1997), no. 4, 685–698.
- [AC04] I. Arzhantsev, O. Chuvashova. Classification of affine homogeneous spaces of complexity one. (Russian) Mat. Sb. 195 (2004), no. 6, 3–20; translation in Sb. Math. 195 (2004), no. 5-6, 765–782.
- [BM13] V. Batyrev, A. Moreau. The arc space of horospherical varieties and motivic integration. Compositio Math. 149 (2013), no. 8, 1327–1352.
- [Bri89] M. Brion. Groupe de Picard et nombre charactéristiques des variétés sphériques. Duke Math. J. 58 (1989), no.2, 397–424.
- [Bri91] M. Brion. Sur la géométrie des variétés sphériques. Comment. Math. Helvetici 66 (1991) 237–262.
- [Bri93] M. Brion. Variétés sphériques et théorie de Mori. Duke Math. J. 72 (1993), no.2, 369–404.
- [FZ03] H. Flenner, M. Zaidenberg. Normal affine surfaces with C<sup>\*</sup>-actions. Osaka J. Math. 40 (2003), no. 4, 981–1009.
- [EGA IV] A. Grothendieck. Éléments de géométrie algébrique. IV. Étude locale des schémas et des morphismes de schémas IV. Inst. Hautes Études Sci. Publ. Math. No. 32 1967.
- [Har77] R. Hartshorne. Algebraic geometry. Graduate Texts in Mathematics, No. 52. Springer-Verlag, New York-Heidelberg, 1977.
- [Hur11] M. Huruguen. Toric varieties and spherical embeddings over an arbitrary field. J. Algebra 342 (2011), 212–234.
- [Jan87] J. C. Jantzen, Representations of algebraic groups. Pure and Applied Mathematics, 131. Academic Press, Inc., Boston, MA, 1987.
- [KKMS73] G. Kempf, F. Knudsen, D. Mumford, B. Saint-Donat. *Toroidal embeddings*. I. Lecture Notes in Mathematics, Vol. 339. Springer-Verlag, Berlin-New York, 1973.
- [Kno90] F. Knop. Weylgruppe und Momentabbildung. Invent. Math. 99 (1990), no. 1, 1–23.
- [Kno91] F. Knop. The Luna-Vust theory of spherical embeddings. Proceedings of the Hyderabad Conference on Algebraic Groups (Hyderabad, 1989), 225–249, Manoj Prakashan, Madras, 1991.
- [Kno93] F. Knop. Über Bewertungen, welche unter einer reduktiven Gruppe invariant sind. Math. Ann. 295 (1993), no. 2, 333–363.

- [Kno94] F. Knop. The asymptotic behavior of invariant collective motion. Invent. Math. 116 (1994), no. 1-3, 309–328.
- [Kol97] J. Kollàr. Singularities of pairs. Algebraic geometry–Santa Cruz 1995, 221–287, Proc. Sympos. Pure Math., 62, Part 1, Amer. Math. Soc., Providence, RI, 1997.
- [Lan15] K. Langlois. Polyhedral divisors and torus actions of complexity one over arbitrary fields.
   J. Pure Appl. Algebra 219 (2015), no. 6, 2015–2045.
- [LS13] A. Liendo, H. Suess. Normal singularities with torus actions. Tohoku Math. J. (2) 65 (2013), no. 1, 105–130.
- [LS13b] A. Liendo, H. Suess. Normal singularities with torus actions. arXiv:1005.2462 (preprint).
- [Lun73] D. Luna. Slices étales. Sur les groupes algébriques, pp. 81–105. Bull. Soc. Math. France, Paris, Mémoire 33 Soc. Math. France, Paris, 1973.
- [LV83] D. Luna, Th. Vust. Plongements d'espaces homogènes. Comment. Math. Helv. 58 (1983), no. 2, 186–245.
- [Mos90] L. Moser-Jauslin. Smooth embeddings of SL(2) and PGL(2). J. Algebra 132 (1990), no. 2, 384–405.
- [Mos92] L. Moser-Jauslin. The Chow rings of smooth complete SL(2)-embeddings. Compositio Math. 82 (1992), no. 1, 67–106.
- [Pan92] D.I. Panyushev. Complexity of quasiaffine homogeneous varieties, t-decompositions, and affine homogeneous spaces of complexity 1. Lie groups, their discrete subgroups, and invariant theory, Adv. Sov. Math. 8 (1992), 151–166.
- [Pas06] B. Pasquier. Variétés horosphériques de Fano. Thèse de doctorat, Institut Fourier (2006).
- [Pas08] B. Pasquier. Variétés horosphériques de Fano. Bull. Soc. Math. France 136 (2008), no. 2, 195–225.
- [Pau81] F. Pauer. Normale Einbettungen von G/U. Math. Ann. 257 (1981), no. 3, 371-396.
- [Pau83] F. Pauer. Glatte Einbettungen von G/U. Math. Ann. 262 (1983), no. 3, 421-429.
- [Per14] N. Perrin. On the geometry of spherical varieties. Transform. Groups 19 (2014), no. 1, 171–223.
- [PS11] L. Petersen, H. Suess. Torus invariant divisors. Israel J. Math. 182 (2011), 481–504.
- [Pez10] G. Pezzini. Lectures on spherical and wonderful varieties. Les cours du C.I.R.M. Actions hamiltoniennes: invariants et classifications. 1 (2010), no. 1, 33–53.
- [PV72] V. L. Popov, E. B. Vinberg. On a class of quasihomogeneous affine varieties. Math. URSS-Izv. 6 (1972), no. 4, 743–758.
- [Pop73] V. Popov. Quasihomogeneous affine algebraic varieties of the group SL(2). (Russian) Izv. Akad. Nauk SSSR Ser. Mat 37 (1973), 792–832.
- [Spr98] T.A. Springer. Linear algebraic groups. Second edition. Birkhäuser, 1998.
- [ST99] M. Strauch, Y. Tschinkel. Height zeta functions of toric bundles over flag varieties. Selecta Math. (N.S.) 5 (1999), no. 3, 325–396.
- [Sum74] H. Sumihiro. Equivariant completion. J. Math. Kyoto Univ. 14 (1974), 1-28.
- [Tim97] D. A. Timashëv. Classification of G-manifolds of complexity 1. (Russian) Izv. Ross. Akad. Nauk Ser. Mat. 61 (1997), no. 2, 127–162; translation in Izv. Math. 61 (1997), no. 2, 363–397.
- [Tim00] D. A. Timashëv. *Cartier divisors and geometry of normal G-varieties*. Transform. Groups 5 (2000), no. 2, 181–204.
- [Tim08] D.A. Timashëv. Torus actions of complexity one. Toric topology, 349–364, Contemp. Math., 460, Amer. Math. Soc., Providence, RI, 2008.
- [Tim11] D. A. Timashëv. Homogeneous spaces and equivariant embeddings. Encyclopaedia of Mathematical Sciences, 138. Invariant Theory and Algebraic Transformation Groups, 8. Springer, Heidelberg, 2011.
- [Vin86] E. B. Vinberg. Complexity of actions of reductive groups. (Russian) Funktsional. Anal. i Prilozhen. 20 (1986), no. 1, 1–13, 96.
- [War82] B. Wargane, *Détermination des valuations invariantes de* SL(3)/T. Thèse, Grenoble, 1982.

INSTITUTO DE CIENCIAS MATEMATICAS (ICMAT), CAMPUS CANTOBLANCO, MADRID, SPAIN *E-mail address:* langlois.kevin180gmail.com

Fachbereich Physik, Mathematik und Informatik, Johannes Gutenberg – Universität Mainz, Mainz, Germany

*E-mail address:* rterpere@uni-mainz.de